Scio Diamond Technology Corp Form ARS October 09, 2014

This document was generated as part of a paper submission. Please reference the Document Control Number 14008317 for access to the original document.

",DN="1",CHK=928591,FOLIO='18',FILE='DISK124:[06HOU4.06HOU1074]DG1074A.;21',USER='KBLACKW',CD=';6-A

conduct any physical inspection of Stewart & Stevenson's properties or facilities. With respect to the financial forecast information furnished to or discussed with Merrill Lynch by Stewart & Stevenson, Merrill Lynch assumed that it was reasonably prepared and reflected the best currently available estimates and judgment of Stewart & Stevenson's management as to the expected future financial performance of Stewart & Stevenson. Merrill Lynch also assumed that the final form of the merger agreement would be substantially similar to the draft reviewed by it on February 26, 2006.

Merrill Lynch's opinion was necessarily based upon market, economic and other conditions as they existed and could be evaluated on, and on the information made available to Merrill Lynch as of, the date of its opinion.

The following is a summary of the material financial and comparative analyses performed by Merrill Lynch that were presented to Stewart & Stevenson's Board of Directors in connection with its opinion. The financial analyses summarized below include information presented in tabular format. In order to fully understand Merrill Lynch's analyses, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the analyses. Considering the data described below without considering the full narrative description of the financial analyses, including the methodologies and assumptions underlying the analyses, could create a misleading or incomplete view of Merrill Lynch's analyses. The analyses summarized below were presented to Stewart & Stevenson's Board of Directors on the basis of consideration of \$34.75 in cash per share of Stewart & Stevenson common stock. As a result of further discussions with Armor Holdings the merger consideration was subsequently increased to \$35.00 in cash per share of Stewart & Stevenson common stock.

Historical Trading Performance

Using publicly available information, Merrill Lynch reviewed historical trading prices and trading volume data for Stewart & Stevenson common stock during the period from September 15, 2003 until February 24, 2006, and compared the relative share price performance of Stewart & Stevenson during that period to the share price performance of an index consisting of the following companies selected by Merrill Lynch, which are referred to below as the "Defense Equipment Companies":

Alliant Techsystems Inc.Armor Holdings, Inc.General Dynamics CorporationL-3 Communications Holdings, Inc.Lockheed Martin Corporation

Oshkosh Truck Corporation

Merrill Lynch noted the following:

Stock Price Performance Since	Stewart & Stevenson common stock	Defense Equipment Companies
September 15, 2003	59.4%	85.6%
February 25, 2004	77.8%	68.5%
February 25, 2005	34.4%	41.9%
Start of Solicitation Process (November 16, 2005)	24.9%	18.8%

Enterprise Value

Based upon the offer price at which the analyses were presented of \$34.75 per share of Stewart & Stevenson common stock, approximately 30.5 million fully diluted shares of Stewart & Stevenson common stock outstanding, and approximately \$301.7 million of net cash, Merrill Lynch noted that the offer implied an enterprise value for Stewart & Stevenson of approximately \$756.5 million.

Offer Premium

Using publicly available information, Merrill Lynch compared the offer price of \$34.75 per share of Stewart & Stevenson common stock to the historical trading prices for Stewart & Stevenson common stock over various periods ending February 24, 2006. These comparisons were calculated based on actual historical trading prices of Stewart & Stevenson common stock during such periods, as well as based on the implied share price of Stewart & Stevenson common stock assuming that no premium is allocated to Stewart & Stevenson's net cash balance.

Merrill Lynch noted the following offer premia:

Time Period (ending February 24, 2006)	Premium based on actual trading prices of Stewart & Stevenson common stock	Premium based on implied value of common stock assuming no allocation to Stewart & Stevenson's net cash balance
One Day	28.2%	44.7%
30-Day	35.1%	57.4%
90-Day	50.4%	88.7%
180-Day	49.3%	86.5%
Since Start of Solicitation Process		
(November 16, 2005)	52.5%	93.5%
52-Week High	27.6%	43.6%

Comparable Acquisition Analysis

Using publicly available information and financial forecasts provided by Stewart & Stevenson's management, Merrill Lynch compared certain financial and operating information and ratios implied by the merger with comparable financial and operating information based on five transactions in the defense equipment manufacturing industry. The transactions considered and the month and year each transaction was announced were as follows:

Date Announced
August 1997
December 2002
June 2004
March 2005
September 2005

Using publicly available information for each of these transactions, Merrill Lynch reviewed:

enterprise value as a multiple of the latest-twelve-month earnings before interest, taxes, depreciation and amortization immediately preceding announcement of the transaction, which is referred to below as "Enterprise Value/LTM EBITDA"; and

enterprise value as a multiple of the estimated earnings before interest, taxes, depreciation and amortization for the twelve-month period following the announcement of the transaction, which is referred to below as "Enterprise Value/Forward EBITDA."

The analysis indicated the following:

Benchmark	Maximum	Mean	Minimum
Enterprise Value/LTM EBITDA	16.7x	11.8x	8.5x
Enterprise Value/Forward EBITDA	15.5x	10.4x	8.0x

Using a reference range of 9.5x to 12.5x Stewart & Stevenson's latest-twelve-months EBITDA, this analysis indicated a range of implied values per share of Stewart & Stevenson common stock of

approximately \$22.00 to \$25.75, and using a reference range of 8.5x to 10.5x Stewart & Stevenson's 2006 estimated EBITDA, this analysis indicated a range of implied values per share of Stewart & Stevenson common stock of approximately \$24.50 to \$27.75 (in each case based upon approximately 30.2 million fully diluted shares of Stewart & Stevenson common stock outstanding, and \$301.7 million of net cash), compared to the offer price of \$34.75 in cash per share of Stewart & Stevenson common stock.

Comparable Trading Analysis

Using publicly available information and financial forecasts provided by Stewart & Stevenson's management, Merrill Lynch compared certain financial and operating information and ratios for Stewart & Stevenson with corresponding financial and operating information for the Defense Equipment Companies.

Using publicly available information for each of these companies, Merrill Lynch reviewed:

enterprise value as a multiple of the latest-twelve-month earnings before interest, taxes, depreciation and amortization immediately preceding announcement of the transaction, which is referred to below as "Enterprise Value/LTM EBITDA"; and

enterprise value as a multiple of the estimated earnings before interest, taxes, depreciation and amortization for the twelve-month period following the announcement of the transaction, which is referred to below as "Enterprise Value/Forward EBITDA."

The analysis indicated the following:

Benchmark	Maximum	Mean	Minimum	
Enterprise Value/LTM EBITDA	14.3x	10.6x	7.9x	
Enterprise Value/Forward EBITDA	10.7x	9.1x	7.2x	

Using a reference range of 9.5x to 11.5x Stewart & Stevenson's latest twelve-months EBITDA, this analysis indicated a range of implied values per share of Stewart & Stevenson common stock of approximately \$22.00 to \$24.50, and using a reference range of 8.0x to 10.0x Stewart & Stevenson's 2006 estimated EBITDA, this analysis indicated a range of implied values per share of Stewart & Stevenson common stock of approximately 30.2 million fully diluted shares of Stewart & Stevenson common stock outstanding, and \$301.7 million of net cash), compared to the offer price of \$34.75 in cash per share of Stewart & Stevenson common stock.

Discounted Cash Flow Analysis

Using financial forecasts provided by Stewart & Stevenson management, Merrill Lynch performed a discounted cash flow analysis for Stewart & Stevenson for fiscal years 2006 through 2010, inclusive, using discount rates ranging from 10.0% to 12.0% and perpetual growth rates of free cash flow after fiscal year 2010 ranging from 0.5% to 2.0%. This analysis indicated a range of implied values per share of Stewart & Stevenson common stock of approximately \$27.50 to \$34.25, compared to the offer price of \$34.75 per share of Stewart & Stevenson common stock. These ranges of implied values per share of Stewart & Stevenson common stock were based in each case on approximately 30.2 million fully diluted shares of Stewart & Stevenson common stock outstanding, and \$301.7 million of net cash.

The summary set forth above summarizes the material analyses performed by Merrill Lynch but does not purport to be a complete description of the analyses performed by Merrill Lynch in arriving at its opinion. The preparation of a fairness opinion is a complex process and is not necessarily susceptible to partial or summary description. Accordingly, Merrill Lynch believes that its analyses must be considered as a whole and that selecting portions of its analyses and the factors considered by

Merrill Lynch, without considering all analyses and factors, could create an incomplete view of the processes underlying the Merrill Lynch opinion. Merrill Lynch did not assign relative weights to any of its analyses in preparing its opinion. The matters considered by Merrill Lynch in its analyses were based on numerous macroeconomic, operating and financial assumptions with respect to industry performance, general business and economic conditions and other matters, many of which are beyond Stewart & Stevenson's and Merrill Lynch's control, and involve the application of complex methodologies and educated judgments. In addition, no company utilized as a comparison in the analyses described above is identical to Stewart & Stevenson, and none of the transactions utilized as a comparison is identical to the merger.

Stewart & Stevenson's Board of Directors selected Merrill Lynch to deliver its opinion because of Merrill Lynch's reputation as an internationally recognized investment banking firm with substantial experience in transactions similar to the merger and because Merrill Lynch is familiar with Stewart & Stevenson and its business. As part of Merrill Lynch's investment banking business, Merrill Lynch is continually engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, leveraged buyouts, negotiated underwritings, secondary distributions of listed and unlisted securities and private placements.

Merrill Lynch acted as Stewart & Stevenson's financial advisor in connection with the merger and will receive customary fees for its services, a significant portion of which is payable upon the consummation of the merger. In addition, Stewart & Stevenson has agreed to indemnify Merrill Lynch for certain liabilities arising out of its engagement. Merrill Lynch has, in the past, provided financial advisory and financing services to Stewart & Stevenson, including financial advisory services to Stewart & Stevenson in connection with the sale of Stewart & Stevenson's Power Products Division, and financial advisory and financing services to Armor Holdings, and may continue to do so and has received, and may receive, fees for the rendering of such services. In addition, in the ordinary course of business, Merrill Lynch may actively trade Stewart & Stevenson common stock, as well as securities of Armor Holdings, for its own account and for the accounts of its customers and, accordingly, Merrill Lynch may at any time hold a long or short position in such securities.

Effects of the Merger

If the merger is approved by our shareholders and the other conditions to the closing of the merger are either satisfied or waived, Acquisition Corp. will be merged with and into us, and we will be the surviving corporation. After the merger, Armor Holdings will own all of our capital stock.

When the merger is completed, each share of our common stock issued and outstanding immediately prior to the effective time of the merger (other than shares owned or held by us, Armor Holdings, Acquisition Corp. and/or our respective subsidiaries and shares held by shareholders who validly exercise and perfect appraisal rights) will be converted into the right to receive \$35.00 in cash. In exchange for the execution and delivery of an option cancellation agreement, each outstanding option will be canceled and converted into the right to receive an amount in cash determined by multiplying (1) the excess, if any, of \$35.00 over the per share exercise price of the option and (2) the number of shares of common stock subject to the option, net of any applicable withholding taxes and without interest.

At the effective time of the merger, our shareholders will cease to have ownership interests in us or rights as our shareholders. Therefore, you will not participate in any of our future earnings or growth and will not benefit from any of our appreciation in value.

Our common stock is currently registered under the Exchange Act and is quoted on the New York Stock Exchange under the symbol "SVC". As a result of the merger, we will be a privately held corporation, and there will be no public market for our common stock. After the merger, our common

stock will cease to be listed on the New York Stock Exchange. In addition, registration of our common stock under the Exchange Act will be terminated.

If any condition to the merger is not satisfied or waived, including the necessary regulatory approvals, the merger will not be consummated. In that event, you will not receive any cash or other consideration as a result of these transactions.

Risks that the Merger Will Not Be Completed

Completion of the merger is subject to various risks, including, but not limited to, the following:

that the merger agreement and the merger will not be approved and adopted by the holders of at least two-thirds of the outstanding shares of our common stock entitled to vote;

that we will experience a material adverse effect (as defined in the merger agreement and as defined below);

the nature and ability of Armor Holdings and Acquisition Corp. to consummate the merger, including satisfying the conditions thereto;

that the parties will not have performed in all material respects their obligations contained in the merger agreement at or before the effective time of the merger;

that we will not secure required governmental consents to, and approvals for, the merger;

that the representations and warranties made by the parties in the merger agreement will not be true and correct to the extent required in the merger agreement immediately before the effective time of the merger;

that there may be brought or pending any suit, action or proceeding by any United States Federal governmental entity seeking to restrain or prohibit the consummation of the merger; and

that there shall have been enacted or issued by any governmental entity any injunction, order or other legal restraint restricting or prohibiting consummation of the merger.

As used in the merger agreement, a "material adverse effect" or "material adverse change" with respect to us means any effect, change or development that individually, or together with other effects, changes or developments, is (or would be) material and adverse to:

our and our subsidiaries' financial condition, results of operations, assets, liabilities or business, taken as a whole, or

our ability to consummate the transactions contemplated by the merger agreement, excluding in each case the impact of:

changes in laws of general applicability or changes in laws of specific applicability to businesses engaged in contracting with governmental authorities as to the supply of products and services for use in, or related to, military activities ("Defense Businesses") and not disproportionally and materially impacting us, or changes in the interpretation thereof by governmental authorities,

changes in general economic, financial market or political conditions or changes in economic conditions affecting Defense Businesses (including any outbreak or escalation of hostilities or war or any act of terrorism), or

changes resulting from the announcement or the existence of, or compliance with, the merger agreement and the transactions contemplated thereby.

As a result of various risks to the completion of the merger, there can be no assurance that the merger will be completed even if the requisite shareholder approval is obtained. If our shareholders do not approve and adopt the merger agreement and the merger or if the merger is not completed for any reason, it could have an adverse effect on:

our relationships with customers and suppliers;

employee morale and the potential loss of executives and other key employees; and

our sales, operating results and stock price.

Interests of Specific Persons in the Merger

In considering the recommendation of our Board of Directors, you should be aware that certain members of our management team and Board of Directors, including our Chief Executive Officer and President, Max L. Lukens, have interests in the merger that are different from, or in addition to, your interests as a shareholder. These interests include compensatory payments to be paid pursuant to an employment agreement with Mr. Lukens, severance agreements with each of our five executive officers and the cancellation and termination of stock options held by our five executive officers and six non-officer directors. Our Board of Directors was aware of, and considered, the interests of our directors and executive officers in approving the merger agreement and the merger. For the reasons, among others, described below, these additional interests did not affect the ability of our Board of Directors to exercise its judgment solely for the benefit of us and our shareholders:

our Board of Directors was aware of these additional interests and the Board's decision to approve the merger agreement and the merger was made with full knowledge of the existence of these additional interests and their terms;

the executive officers' severance agreements were recommended by non-employee directors more than 18 months prior to the execution of the merger agreement (except that our original severance agreement entered into with Mr. Lukens effective February 1, 2004, was replaced in December 2005 in connection with its scheduled expiration in February 1, 2006 and our severance agreement with Mr. L. Scott Biar, our Chief Financial Officer, was entered into in June 2005) and all were recommended for the purpose of ensuring that our executive officers would continue to devote their services for the best interests of us and our shareholders and not be distracted by any potential change in control;

our directors believed the executive officers' severance agreements to be prudent and in the best interests of us and our shareholders;

these additional interests were not implemented for the purpose of creating a deterrent to any takeover proposal;

all of the stock options held by our executive officers and directors were granted in the normal course of business in accordance with our past practices and not in contemplation of the proposed transaction with Armor Holdings or any other party; and

six of our seven directors are non-employee directors, and the only compensation that such non-employee directors will receive in connection with the merger (other than cash for any shares of our common stock held by such director) results from the cash-out of their stock options, none of which had been granted in connection with the merger agreement or the merger.

Employment and Severance Agreements

In February 2004, we entered into an employment agreement and a severance agreement with Mr. Lukens that provided for, among other things, in the case of his employment agreement, a cash payment to him if his employment was terminated by him for good reason or by us without cause, and in the case of the severance agreement, a cash payment to him if his employment was terminated by him for good reason or by us or our successor without cause following a change in control of our company. As a result of the scheduled expiration of these agreements in February 2006, we entered into a new employment agreement and a new severance agreement with Mr. Lukens in December 2005. Under Mr. Lukens' employment agreement and his severance agreement, if his employment is terminated under specified circumstances in connection with a change of control, such as the proposed merger, then he will be entitled to all of the following:

all amounts due to him as of the date of termination;

a lump sum cash payment equal to \$3,000,000;

continuation of life, disability, accident and health insurance benefits and all perquisites until February 1, 2010; and

an additional amount (a "gross-up" payment) to cover any excise taxes that may be imposed under the "golden parachute" rules on payments and benefits received in connection with a change in control, so that Mr. Lukens would receive the total payment initially allocated by us.

In July 2004, we entered into severance agreements with Carl B. King, our Senior Vice President and General Counsel, Dennis M. Dellinger, Vice President, and Stephen A. Hines, Vice President. In June 2005, we entered into a severance agreement with L. Scott Biar, our current Chief Financial Officer. If the employment of any of Messrs. King, Dellinger, Hines or Biar is terminated under specified circumstances in connection with a change of control, such as the proposed merger, then such person will be entitled to all of the following:

a lump sum cash payment equal to two times the sum of the executive's base salary and the average cash value of annual bonus (whether paid in cash or stock) earned by the executive in the three fiscal years prior to the date of termination (as defined in the severance agreement), provided the executive has participated in our annual bonus or incentive plan for that period. If not, then the average is calculated using the lesser number of bonuses actually earned;

continuation of life, disability, accident and health insurance benefits and perquisites for an additional three years;

notwithstanding any provision of our incentive compensation plan, a lump sum cash payment equal to the sum of any unpaid incentive compensation allocated or awarded to the executive and a pro rata portion to the date of termination of the aggregate value of all contingent incentive compensation awards;

outplacement services suitable to executive's position for a period of one year;

a nonforfeitable interest in the executive's benefits accrued under our supplemental retirement plan; and

an additional amount (a "gross-up" payment) to cover any excise taxes that may be imposed under the "golden parachute" rules on payments and benefits received in connection with a change in control, so that the executive would receive the total payment initially allocated by us.

We entered into these severance agreements as a result of the determination of our Board of Directors and compensation committee that it was imperative for us to be able to rely upon these individuals to continue in their respective positions with us without concern that such employees might be distracted by the personal uncertainties and risks created by any proposed change of control of our company.

The table below sets forth the base salary and severance and estimated other payments for each executive officer who has an employment agreement or severance agreement with us:

Executive Officers	Ba	se Salary	Estimated Severance Payment(1)
Max L. Lukens	\$	750,000	\$ 3,211,000(2)
Carl B. King	\$	325,000	\$ 1,298,000(3)
Dennis M. Dellinger	\$	290,000	\$ 1,204,000(3)
L. Scott Biar	\$	225,000	\$ 698,000(3)
Stephen A. Hines	\$	200,000	\$ 677,000(3)
Total			\$ 7,088,000

(1)

The amount payable under each severance agreement is subject to variables and can vary depending upon the circumstances then in existence at the time such amounts become payable. Therefore, in order to quantify the amount payable under each severance agreement, several assumptions must be made, including assuming that the merger is consummated on May 9, 2006 and that a qualifying termination occurs on that date. Excludes cash out of options, payment for shares owned, health and welfare benefits and any excise tax gross-up payments or any other tax-gross up payments that may be payable to executives pursuant to the severance agreements.

(2)

Consists of the sum of (i) specified severance payment and bonus pursuant to Severance Agreement dated December 23, 2005 and (ii) the cash payment of benefits accured under our supplemental retirement plan.

(3)

Consists of the sum of (i) two times the executive's base salary, (ii) two times the average cash value of annual bonus earned by the executive in respect of fiscal year ended January 31, 2004, 2005 and 2006, (iii) a pro rata portion of incentive compensation earned during fiscal year ended January 31, 2006, assuming the achievement, at expected value target level, of performance goals and (iv) the cash payment of benefits accrued under our supplemental retirement plan.

One of our former executive officers entered into a termination and severance agreement with us dated January 25, 2006, related to his resignation as an executive officer on January 25, 2006 and his resignation as an employee effective March 31, 2006, in connection with the sales of our Engineered Products and Power Products Divisions on January 25, 2006. The agreement provided for the payment of supplemental compensation to the executive in the event of a change of control of our company prior to December 31, 2006. Pursuant to that agreement, the executive officer will be entitled to receive a lump-sum payment of \$974,000 on the later of one day following the date that is six months from the date of his separation from service or the date of the change in control of our company.

Total Consideration for Our Common Stock

Our directors and executive officers will receive the same per share consideration for their shares of our common stock in the merger as all of our other shareholders. The table below sets forth the

estimated aggregate amount of consideration that will be received in the merger by our directors and executive officers on account of their ownership of our common stock as of April 5, 2006:

Directors and Executive Officers	Number of Shares of Our Common Stock Owned	Amount of Cash to be Received for Our Common Stock in the Merger
Max L. Lukens	31,266	\$ 1,094,310
Carl B. King		
Dennis M. Dellinger	1,762	\$ 61,670
L. Scott Biar		
Stephen A. Hines	2,620	\$ 91,700
Howard Wolf	21,776	\$ 762,160
Monroe M. Luther	9,755	\$ 341,425
Charles R. Ofner	4,755	\$ 166,425
Charles S. Ream	1,285	\$ 44,975
Robert S. Sullivan	4,787	\$ 167,545
James M. Tidwell	2,565	\$ 89,775
Total	80,571	\$ 2,819,985

Stock Options

If our shareholders approve the merger agreement and the merger, and the merger is completed, in exchange for the execution and delivery of an option cancellation agreement, each outstanding option to purchase our common stock will be cancelled and converted into the right to receive (1) the excess, if any, of \$35.00 over the per share exercise price of the option multiplied by (2) the number of shares of common stock subject to the option, net of any applicable withholding taxes and without interest. The following table sets forth the number of shares of our common stock subject to stock options that are owned by our executive officers and directors as of April 5, 2006, and the amount, assuming no option exercises prior to the consummation of the merger, that each executive officer and director will receive in exchange for the cancellation and termination of those options in connection with the merger:

Directors and Executive Officers	Number of Shares of Our Common Stock Subject to Options	Payment Upon Completion Of the Merger and Execution of an Option Cancellation Agreement(1)	
Max L. Lukens	581,000	\$	10,998,950
Carl B. King	213,500	\$	4,360,750
Dennis M. Dellinger	180,000	\$	3,521,600
L. Scott Biar	27,500	\$	499,150
Stephen A. Hines	64,300	\$	1,301,885
Howard Wolf	54,000	\$	938,463
Monroe M. Luther	26,000	\$	411,963
Charles R. Ofner	26,000	\$	411,963
Charles S. Ream	10,000	\$	158,650
Robert S. Sullivan	29,000	\$	443,150
James M. Tidwell	10,000	\$	154,350
Total	1,221,300	\$	23,200,874

(1)

Determined by multiplying (1) the excess, if any, of \$35.00 over the per share exercise price of the option and (2) the number of shares of our common stock subject to the option.

Directors and Officers

It is currently expected that none of our directors or executive officers will become a director or officer of Armor Holdings or Acquisition Corp. Armor Holdings and Acquisition Corp. currently have no arrangements or agreements with any of our officers following completion of the merger.

Employee Compensation and Benefits

From and after the effective time of the merger until the date six months after the closing of the transactions contemplated by the merger agreement, Armor Holdings has agreed to provide our employees and former employees (including executive officers) taken as a whole with employee benefits and compensation plans, programs and arrangements (including base salary and annual bonus opportunities, but specifically excluding equity grants and any supplemental executive retirement plan benefits) no less favorable, in the aggregate, than those provided by us and our subsidiaries to such persons immediately prior to the effective time of the merger. After such six month period, Armor Holdings has agreed to provide our employees and former employees (including executive officers) taken as a whole with employee benefits and compensation plans, programs and arrangements and severance benefits that are no less favorable, in the aggregate, to those provided to similarly situated employees of Armor Holdings and its subsidiaries. In addition, during the 24 months after the effective time of the merger, any such person whose employment is terminated "without cause" (as defined in such policy) shall be entitled to receive severance payments and benefits under our severance policy disclosed to Armor Holdings.

We and Armor Holdings have also agreed that, subject to Armor Holdings' right to propose modifications to conform to the terms and conditions of its bonus plans for similarly situated employees, each of our executive officers will be eligible to participate in a bonus plan for the fiscal year ending January 31, 2007 based upon targets and goals substantially similar to those established for our fiscal year ending January 31, 2007. The proposed terms of the bonus plan provide for a payment to the executive officers (other than Mr. Lukens) if their employment is terminated prior to October 31, 2006 and the year-to-date performance metric through the most recently completed month meets or exceeds the year-to-date forecasted results computed from the quarterly performance metric targets provided by us to Armor Holdings.

Directors' and Officers' Indemnification and Insurance

The merger agreement provides that the indemnification provisions in the articles of incorporation and bylaws of the surviving corporation (which shall be the same provisions as exist in our restated articles of incorporation and amended and restated bylaws, as amended, immediately prior to the completion of the merger) shall not be amended, repealed or otherwise modified for six years after the completion of the merger in any manner that would adversely affect the rights thereunder of the individuals who were directors, officers and employees at the effective time of the merger, unless counsel advises that such modification is legally required.

In the merger agreement, Armor Holdings and Acquisition Corp. have agreed to indemnify, defend and hold harmless our and our subsidiaries' current and former directors, officers and employees against all liability arising out of actions or omissions before or at the effective time of the merger to the fullest extent permitted by law and our restated articles of incorporation and amended and restated bylaws, as amended.

Under the merger agreement, Armor Holdings has further agreed that, for a period of six years after the effective time of the merger, it will cause to be maintained in effect director's and officer's liability insurance to reimburse the current and former officers and directors with respect to claims against them arising from facts or events occurring before or at the effective time of the merger. Armor

Holdings is obligated to ensure such insurance contains at least the same coverage and amounts of coverage and will contain terms and conditions no less advantageous to the indemnified individuals then those provided in the director's and officer's liability insurance currently provided by us; however, Armor Holdings is not obligated to pay a premium for such coverage in excess of 200% of the current premium, in which event the obligation shall be only to provide the maximum coverage available for a premium not in excess of such amount. The merger agreement further provides that the foregoing obligations shall be satisfied by our purchase on or before the effective time of the merger of a tail-coverage or run-off insurance policy providing the required coverage for the required term and with a premium that does not exceed 200% of the current premium.

If Armor Holdings or any of its successors or assigns: (1) consolidates with or merges into any other entity and is not the continuing or surviving entity or (2) transfers all or substantially all of its assets to any entity, the surviving or transferee entity shall assume the indemnification and insurance obligations discussed above.

Federal Regulatory Matters

The HSR Act, and the rules and regulations promulgated thereunder, require that each of us and Armor Holdings file notification and report forms with respect to the merger and related transactions with the Antitrust Division of the Department of Justice, or the DOJ, and the Federal Trade Commission, or the FTC. We and Armor Holdings filed the notification and report forms on March 17, 2006 and March 20, 2006, respectively. The parties are required to observe a waiting period after these filings before completing the merger and the transactions contemplated by the merger agreement. The waiting period will expire 30 days after the filings were made unless the DOJ or FTC issues a second request for additional information, in which case the waiting period will expire 30 days after substantial compliance with such request. The required waiting period may be terminated by the FTC or the DOJ before its expiration. The DOJ, the FTC, state antitrust authorities or a private person or entity could seek to enjoin the merger under federal or state antitrust laws at any time before completion of the merger or to compel rescission or divestiture at any time subsequent to the merger. We cannot assure you that a challenge to the merger will not be made by the DOJ, the FTC, state antitrust authorities or a private person or entity or that, if made, Armor Holdings and we would prevail or would not be required to accept various conditions, including divestitures, to complete the merger.

Transactions with Armor Holdings

One of our subsidiaries and a subsidiary of Armor Holdings are parties to

teaming agreements relating to the joint development of the armored cab for the U.S. Army's Family of Medium Tactical Vehicles, which includes the High Mobility Artillery Rocket System, and

purchase orders for the supply by a subsidiary of Armor Holdings to one of our subsidiaries of armoring materials for incorporation into our Low Signature Armored Cabs.

Material U.S. Federal Income Tax Consequences

The following discussion summarizes the material U.S. federal income tax consequences of the merger that are generally applicable to United States holders (as defined below) of our common stock. This discussion is based on currently existing provisions of the Internal Revenue Code of 1986, as amended, referred to as the Code in this proxy statement, existing and proposed Treasury Regulations promulgated under the Code, and current administrative rulings and court decisions, all of which are subject to change, possibly with retroactive effect. This discussion does not address state, local or foreign tax consequences that may be applicable to the parties specified in the first sentence of this paragraph, and such parties should consult their own tax advisors with respect to such consequences.

United States Holders

The following discussion applies only to United States holders of our common stock who hold such shares as capital assets and may not apply to shares of our common stock acquired pursuant to the exercise of employee stock options or other compensation arrangements (and does not apply to the exchange or cancellation of employee stock options, including the receipt of cash therefor), and this discussion does not address tax issues relevant to certain classes of taxpayers who may be subject to special treatment under the Code, such as banks, other financial institutions, insurance companies, tax-exempt investors, regulated investment companies, real estate investment trusts, persons subject to the alternative minimum tax, persons who hold their shares of our common stock as part of a position in a "straddle" or as part of a "hedging" or "conversion" transaction, persons who are deemed to sell their shares of our common stock under the constructive sale provisions of the Code, traders in securities that elect to use a mark-to-market method of accounting for their securities holdings, persons that have a functional currency other than the U.S. dollar, expatriates, S corporations, entities classified as partnerships for U.S. federal income tax purposes or shareholders who hold shares of our common stock as dealers. All such United States holders should consult their own tax advisors concerning the U.S. federal income tax consequences of the merger to their particular situations.

THIS SUMMARY IS NOT A SUBSTITUTE FOR AN INDIVIDUAL ANALYSIS OF THE TAX CONSEQUENCES OF THE MERGER TO YOU. TAX MATTERS ARE VERY COMPLEX AND THE TAX CONSEQUENCES OF THE MERGER TO YOU WILL DEPEND ON THE FACTS OF YOUR PARTICULAR SITUATION. YOU SHOULD CONSULT YOUR TAX ADVISOR FOR A FULL UNDERSTANDING OF THE TAX CONSEQUENCES OF THE MERGER TO YOU, INCLUDING THE FEDERAL, STATE, LOCAL AND FOREIGN TAX CONSEQUENCES OF THE MERGER.

If a partnership (including for this purpose any entity treated as a partnership for U.S. federal income tax consequences) holds shares of our common stock, the tax treatment of a partner will generally depend on the status of the partner and the tax treatment of the partnership. Partnerships and partners in partnerships holding shares of our common stock should consult their tax advisors.

For purposes of this discussion, a "United States holder" means a holder that is:

a citizen or resident alien of the United States,

a corporation (or other entity treated as an association taxable as a corporation for U.S. federal income tax purposes) created or organized in or under the laws of the United States or any state,

an estate, the income of which is subject to U.S. federal income taxation regardless of its source, or

a trust if:

a U.S. court is able to exercise primary supervision over the administration of the trust and one or more U.S. persons have authority to control all substantial decisions of the trust, or

the trust has a valid election in effect under applicable Treasury Regulations to be treated as a U.S. person.

In general, United States holders of shares of our common stock who receive cash in exchange for their shares pursuant to the merger should recognize gain or loss for U.S. federal income tax purposes equal to the difference, if any, between the amount of cash received and the adjusted tax basis in their shares of our common stock. If a shareholder holds our common stock as a capital asset, the gain or loss should generally be a capital gain or loss. If the shareholder has held the shares for more than one year, the gain or loss should generally be a long term gain or loss. The deductibility of capital losses is subject to limitations.

Holders of our common stock are entitled to appraisal rights under Texas law in connection with the merger. If a United States holder receives cash pursuant to the exercise of appraisal rights, that United States holder generally will recognize gain or loss measured by the difference between the amount of cash received and the adjusted tax basis in their shares of our common stock. This gain should be long-term capital gain or loss if the United States holder held our common stock for more than one year. Any holder of our common stock that plans to exercise appraisal rights in connection with the merger is urged to consult a tax advisor to determine the related tax consequences.

U.S. federal income tax law requires that a holder of our common stock provide the disbursing agent with his or her correct taxpayer identification number, which is, in the case of a United States holder who is an individual, a social security number, or, in the alternative, establish a basis for exemption from backup withholding. Exempt holders, including, among others, corporations and some foreign individuals, are not subject to backup withholding and reporting requirements. If the correct taxpayer identification number or an adequate basis for exemption is not provided, a holder will be subject to backup withholding on any reportable payment. Any amounts withheld under the backup withholding rules from a payment to a United States holder will be allowed as a credit against that United States holder's U.S. federal income tax and may entitle the United States holder to a refund, if the required information is furnished to the Internal Revenue Service.

To prevent backup withholding, each United States holder of our common stock must complete the Substitute Form W-9 which will be provided by the disbursing agent with the transmittal letter and certify under penalties of perjury that:

the taxpayer identification number provided is correct or that the United States holder is awaiting a taxpayer identification number, and

the United States holder is not subject to backup withholding because:

the United States holder is exempt from backup withholding,

the United States holder has not been notified by the Internal Revenue Service that such holder is subject to backup withholding as a result of the failure to report all interest or dividends, or

the Internal Revenue Service has notified the United States holder that such holder is no longer subject to backup withholding.

The Substitute Form W-9 must be completed, signed and returned to the disbursing agent.

Appraisal Rights

THE FOLLOWING DISCUSSION IS A SUMMARY OF THE MATERIAL STATUTORY PROCEDURES TO BE FOLLOWED BY A HOLDER OF OUR COMMON STOCK IN ORDER TO DISSENT FROM THE MERGER AND PERFECT DISSENTERS' RIGHTS OF APPRAISAL. IF YOU WANT TO EXERCISE APPRAISAL RIGHTS, YOU SHOULD REVIEW CAREFULLY ARTICLES 5.11, 5.12 AND 5.13 OF THE TBCA, WHICH ARE ATTACHED AS APPENDIX C TO THIS PROXY STATEMENT. YOU ARE URGED TO CONSULT A LEGAL ADVISOR BEFORE ELECTING OR ATTEMPTING TO EXERCISE THESE RIGHTS. THE FAILURE TO PRECISELY FOLLOW ALL NECESSARY LEGAL REQUIREMENTS MAY RESULT IN THE LOSS OF APPRAISAL RIGHTS. THIS DESCRIPTION IS NOT COMPLETE AND IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO THE FULL TEXT OF ARTICLES 5.11, 5.12 AND 5.13 OF THE TBCA. SHAREHOLDERS SEEKING TO EXERCISE APPRAISAL RIGHTS MUST STRICTLY COMPLY WITH THESE PROVISIONS.

Shareholders as of the record date who follow the procedures set forth in Articles 5.11, 5.12 and 5.13 of the TBCA will be entitled to demand, if the merger is consummated, that we purchase their

shares of our common stock for an amount in cash equal to the "fair value" of their shares. Under the TBCA, fair value of shares for the purposes of exercise of appraisal rights is defined as the value of the shares as of the day immediately prior to the date of the special meeting, excluding any appreciation or depreciation in the value of the shares in anticipation of the proposed merger. This value may differ from the value of the consideration that you would otherwise receive in the merger.

How to Exercise and Perfect Your Right to Dissent

In order to be eligible to exercise your right to dissent from the merger and to receive the fair value of your shares of our common stock as of the day immediately preceding the special meeting, excluding any appreciation or depreciation in anticipation of the proposed merger, you must provide us with a written objection to the merger prior to the special meeting. The written objection must state that you intend to exercise your right to dissent if the merger is consummated and provide an address to which a notice about the outcome of the vote on the merger may be sent. Neither a proxy nor a vote against the plan of merger is sufficient to constitute a written objection as required under the TBCA. *We must receive your written objection to the merger prior to the special meeting*.

Any written objection with notice of intent to exercise the right of dissent should be addressed as follows:

Stewart & Stevenson Services, Inc. c/o Carl B. King Senior Vice President, General Counsel and Secretary 2707 North Loop West P. O. Box 1637 Houston, Texas 77251-1637

In order to exercise appraisal rights properly, you must not vote, whether by proxy or in person, in favor of the merger agreement and the transactions contemplated thereby. If you execute and return an unmarked proxy, your shares will be voted "FOR" the merger and, as a consequence, you will be foreclosed from exercising rights of appraisal as a dissenting shareholder.

Your Demand for Payment

If the merger is consummated, we will within 10 days after the effective date of the merger deliver or mail to all of our shareholders who have satisfied the foregoing requirements concerning the exercise of appraisal rights a written notice that the merger has been consummated. If you want to exercise your dissenters' rights of appraisal, you must, within 10 days from the delivery or mailing of such notice, send a written demand to us for payment of the fair value of your shares of our common stock. Such written demand must state the number and class of the shares that you owned as of the record date and your estimate of the fair value of the shares. Under Texas law, the fair value of your shares of our common stock for the purpose of exercising dissenters' rights of appraisal will be the value of the shares on the day immediately preceding the special meeting, excluding any appreciation or depreciation in anticipation of the merger. In computing the fair value of your shares, consideration will be given to our value as a going concern without including in the computation of value any payment for a control premium or minority discount other than a discount attributable to the type of share held by you and any limitation placed on the rights and preferences of your shares. If you fail to make such a demand within the 10-day period, you will lose the right to dissent and will be bound by the terms of the merger agreement. In order to preserve your dissenters' rights of appraisal, you must also submit to us your stock certificates, if any, representing your shares of common stock within 20 days after making a demand for payment for notation thereon that such demand has been made. The failure to do so will, at our option, terminate your dissenters' rights of appraisal unless a court of

competent jurisdiction for good and sufficient cause shown directs otherwise. Your written demand and any notice should be sent to us at the address set forth above.

Action Upon Receipt of Your Demand for Payment

Within 20 days after receiving your written demand for payment and estimate of the fair value of your shares of our common stock, we must mail or deliver to you a written notice that either:

accepts the amount declared in the demand and agrees to pay that amount within 90 days after the effective date of the merger and, in the case of shares represented by certificates, upon surrender of duly-endorsed certificate(s) representing your shares of our common stock; or

states our estimate of the fair value of the shares and offers to pay the amount of that estimate within 90 days after the effective date of the merger upon receipt of notice from you within 60 days after the effective date of the merger that you agree to accept our estimate and, in the case of shares represented by certificates, upon surrender of duly endorsed certificate(s) representing your shares of our common stock.

Payment of the Fair Value of Your Shares of Our Common Stock Upon Agreement of an Estimate

If you and we agree upon the fair value of your shares of our common stock within 60 days after the effective date of the merger, we will pay you the amount of the agreed value within 90 days after the effective date of the merger and, in the case of shares represented by certificates, upon receipt of your duly-endorsed stock certificates. Upon payment of the agreed fair value, you will cease to have any interest in such shares.

Commencement of Legal Proceedings if a Demand for Payment Remains Unsettled

If an agreement upon the fair value of your shares of our common stock is not reached between you and us within the 60-day period immediately following the effective date of the merger, then either you or we may, within 60 days after the expiration of the 60-day period following the effective date of the merger, file a petition in any court of competent jurisdiction in Harris County, the county in Texas where our principal office is located, asking for a finding and determination of the fair value of the shares. If filed by a shareholder, service of the petition shall be made upon us and we must within 10 days after service file with the clerk of the court a list with the names and addresses of all shareholders who have demanded payment and not reached agreement as to the fair value. If filed by us, the petition must be accompanied by such a list. The clerk of the court shall give notice to us and all shareholders named on the list of the time and place fixed for the hearing of the petition.

After the hearing of the petition, the court shall determine the shareholders who have complied with the statutory requirements and have become entitled to the valuation of and payment for their shares, and the court shall appoint one or more qualified appraisers to determine the fair value. The appraisers may examine our books and records and must afford the interested parties a reasonable opportunity to submit pertinent evidence as to the value of the shares. The appraisers are to make a determination of the fair value upon such examination as they deem proper. The appraisers will file a report of the value in the office of the clerk of the court, notice of which will be given to the parties in interest. The parties in interest may submit exceptions to the report, which will be heard before the court upon the law and the facts. The court will adjudge the fair value of the shares of the shareholders entitled to payment for their shares and will direct the payment thereof by us, together with interest which will begin to accrue 91 days after the effective date of the merger. The judgment will be payable to the holders of uncertificated shares immediately but to the holders of shares represented by certificates only upon, and simultaneously with, the surrender to the surviving corporation of duly endorsed certificates of those shares. Upon our payment of the judgment, you will cease to have any interest in the shares. In the absence of fraud, the remedy provided by Article 5.12



of the TBCA to a shareholder objecting to the merger is the exclusive remedy for the recovery of the value of such shareholder's shares or money damages with respect to the merger.

The court must allow the appraisers a reasonable fee as court costs, and all court costs will be allotted between the parties in the manner that the court determines to be fair and equitable.

Any shareholder who has demanded payment for that holder's shares may withdraw the demand at any time before payment or before any petition has been filed for valuation by the court. A demand may not be withdrawn after payment or, unless we consent, after such a petition has been filed in court. After a demand has been withdrawn, the shareholder and all persons claiming under the shareholder will be conclusively presumed to have approved the plan of merger and will be bound by its terms.

THE MERGER AGREEMENT

The following is a summary of the material terms of the merger agreement, which is attached to this proxy statement as Appendix A and is incorporated by reference into this proxy statement. This summary does not purport to be complete and may not contain all of the information about the merger agreement that is important to you. To understand the merger more fully, and for a more complete legal description of the merger, you are urged to read carefully the entire proxy statement, including the merger agreement and the other appendices.

The descriptions of the merger agreement in this proxy statement have been included to provide you with information regarding its terms. Except for its status as the contractual document between the parties with respect to the merger, it is not intended to provide factual information about Armor Holdings, Acquisition Corp. or us. The merger agreement contains representations and warranties made by and to Armor Holdings, Acquisition Corp. and us as of specific dates. The representations and warranties were made for purposes of the merger agreement, including qualifications and limitations agreed to by the parties in connection with negotiating the terms of the merger agreement, including qualifications in disclosures exchanged between the parties. In addition, some representations and warranties were made as of a specified date, may be subject to contractual standards of materiality different from those generally applicable to shareholders, and have been made solely for purposes of risk allocation and to provide contractual rights and other remedies to Armor Holdings, Acquisition Corp. and us. You should not rely upon the representations and warranties set forth in the merger agreement as statements of factual information.

Structure of the Merger

The merger agreement provides that, following the approval of the merger agreement by our shareholders and the satisfaction or waiver of the other conditions to the merger, including receipt of required regulatory approvals, Acquisition Corp. will be merged with and into us. We will be the surviving corporation in the merger and will be a wholly-owned subsidiary of Armor Holdings.

Effective Time of the Merger

While we intend to consummate the merger as promptly as practicable subject to receipt of shareholder approval and the satisfaction or waiver of the other conditions to the merger, we cannot specify when, or assure you that, all conditions to the merger will be satisfied or waived.

The merger will become effective upon the issuance of a certificate of merger by the Secretary of State of the State of Texas following the filing of articles of merger and/or a certificate of merger with the Secretary of State of the State of Texas on the closing date of the merger (or at such later time as agreed to by the parties and specified in the articles of merger).

Articles of Incorporation; Bylaws and Directors and Officers of the Surviving Corporation

When the merger becomes effective, our amended and restated articles of incorporation, as in effect immediately before the effective time, will become the articles of incorporation of the surviving corporation, except that the name of the surviving corporation will be changed therein to the name of our company immediately prior to the effective time. The by-laws of Acquisition Corp., as in effect immediately prior to the effective time, will become the by-laws of the surviving corporation, except to the extent they need to be amended to effectuate the indemnification obligations of Armor Holdings and Acquisition Corp. in the merger agreement.

The directors of Acquisition Corp. immediately prior to the effective time of the merger will be the directors of the surviving corporation following the merger until their respective successors are duly elected or appointed and qualified. The officers of the surviving corporation will be as set forth in the

merger agreement, subject to the right of the Board of Directors of the surviving corporation to appoint and replace officers.

Merger Consideration

At the effective time of the merger each issued and outstanding share of our common stock, except those (1) shares owned or held by the us, Armor Holdings or Acquisition Corp. or our respective subsidiaries, and (2) shares held by our shareholders who validly exercise their dissenters' rights under Texas law, will automatically be cancelled and converted into the right to receive \$35.00 in cash, without interest. Shares of our common stock that are outstanding immediately prior to the effective time of the merger and that are held by any dissenting shareholder who properly exercises his or her appraisal rights will not be canceled and converted into the right to receive \$35.00 in cash; rather the dissenting shareholder will be entitled to the fair value of his or her shares in accordance with and subject to Articles 5.11, 5.12 and 5.13 of the TBCA, including compliance by the shareholder with the procedures therein. See "Special Factors Appraisal Rights" on page 31.

At the effective time of the merger, each issued and outstanding share of common stock of Acquisition Corp. will be converted into one fully paid and non-assessable share of common stock, without par value, of the surviving corporation.

Exchange Procedures

At or prior to the effective time, Armor Holdings will deposit with a disbursing agent agreed upon between Armor Holdings and us an amount in cash sufficient to pay the aggregate merger consideration payable to all holders of our common stock and all holders of options to purchase our common stock. Armor Holdings will also make funds available to the disbursing agent from time to time after the effective time as needed to pay the merger consideration. The disbursing agent will deliver to Armor Holdings any cash remaining in its possession, together with any earnings in respect thereof, one year after the effective time of the merger. Thereafter, Armor Holdings will act as the disbursing agent and any former holder of our common stock may look only to Armor Holdings or the surviving corporation for payment of merger consideration to which they may be entitled. None of us, Armor Holdings, Acquisition Corp. or the disbursing agent will be liable to any former shareholder of ours for any amount properly delivered to a public official pursuant to applicable abandoned property, escheat or similar laws.

Within two business days of the effective time of the merger, the disbursing agent will mail to each record holder of shares of our common stock a transmittal letter containing instructions to effect the surrender of the holder's certificates in exchange for payment of the merger consideration. Our shareholders who surrender their stock certificates to the disbursing agent, together with a properly completed and signed transmittal letter and any other documents required by the instructions to the transmittal letter, will receive \$35.00 per share of our common stock surrendered to the disbursing agent, less any applicable withholding taxes.

You should not return your stock certificates representing shares of our common stock with the enclosed proxy card, and you should not forward your stock certificates to the disbursing agent without a transmittal letter. In all cases, the merger consideration will be paid only in accordance with the procedures set forth in the merger agreement and the transmittal letter.

If payment of the merger consideration is to be made to a person other than the person in whose name a share certificate is registered, it will be a condition of payment that the person requesting such payment either pay any applicable taxes required or establish, to the reasonable satisfaction of the disbursing agent, that the tax has been paid or is not applicable.

If any share certificate is lost, stolen or destroyed, the disbursing agent will pay the merger consideration for the number of shares represented by such certificate upon delivery by the person

seeking payment of an affidavit in lieu of the certificate, and if required by the surviving corporation, an indemnity bond in form and substance and with surety reasonably satisfactory to the surviving corporation.

At the effective time of the merger, our stock transfer books will be closed and no further issuances or transfers of our shares will be made.

Treatment of Options

We will terminate our four existing stock option plans immediately prior to the effective time without prejudice to the rights of the holders of options awarded pursuant thereto. At the effective time of the merger, in exchange for the execution and delivery of an option cancellation agreement, each outstanding option to purchase shares of our common stock will be fully vested and exercisable and will be canceled and converted into the right to receive an amount in cash equal to the product of (1) the excess, if any, of \$35.00 over the per share exercise price of the option and (2) the number of shares of the Company's common stock subject to the option, net of any applicable withholding taxes and without interest.

Representations and Warranties

Our Representations and Warranties

We make various representations and warranties in the merger agreement with respect to us and our subsidiaries that are subject, in some cases, to disclosures and specified exceptions and qualifications. Our representations and warranties relate to, among other things:

our due organization, valid existence and good standing;

our capital structure;

our ownership of each of our subsidiaries;

our due organization, valid existence and good standing of our subsidiaries;

the corporate power and authority of us and each of our subsidiaries to own, lease and operate our respective properties and carry on our respective businesses;

our corporate power and authority to enter into, execute and deliver the merger agreement and consummate the transactions contemplated thereby;

our due authorization, execution and delivery of the merger agreement;

the consents, approvals and filings required to enter into the merger agreement or to complete the transactions contemplated by the merger agreement;

the accuracy of our filings with the SEC and compliance with applicable rules and regulations for a specified period of time;

the absence of undisclosed material liabilities that would have a material adverse effect relating to us;

the conduct of our business, and the absence of certain changes in our business, since January 31, 2005;

the absence of material adverse effect relating to us and certain other changes since January 31, 2005;

litigation matters affecting us and our subsidiaries;

the adequacy of the governmental authorizations and permits needed to conduct our business and our compliance with applicable laws and governmental orders;

the absence of defaults or breaches under specified contacts;

tax matters affecting us and our subsidiaries;

labor and employment, employee welfare and benefit plan matters affecting us and our subsidiaries;

the inapplicability of state anti-takeover statutes or regulations to the merger;

our absence of liability for undisclosed broker's fees;

our compliance with the Sarbanes-Oxley Act;

environmental matters affecting us and our subsidiaries;

matters relating to this proxy statement;

the required vote of our shareholders to approve the merger agreement; and

the terms of each confidentiality agreement executed by any person participating in the sale process conducted by us were substantially similar to Armor Holdings' confidentiality agreement with us and there have been no waivers by us under any such other confidentiality agreements.

You should be aware that these representations and warranties made by us to Armor Holdings and Acquisition Corp. may be subject to important limitations, disclosures and qualifications set forth in the merger agreement and the disclosure schedules thereto, and do not purport to be accurate as of the date of this proxy statement or provide factual information about us to our shareholders.

Armor Holdings' Representations and Warranties

Armor Holdings makes various representations and warranties in the merger agreement with respect to it and its subsidiaries that are subject, in some cases, to disclosures and specified exceptions and qualifications. Its representations and warranties relate to, among other things:

its due organization, valid existence and good standing;

its corporate power and authority to own, lease and operate its respective properties and carry on its businesses;

its corporate power and authority to enter into, execute and deliver the merger agreement and consummate the transactions contemplated thereby;

its due authorization, execution and delivery of the merger agreement;

the consents, approvals and filings required by it to enter into the merger agreement or to complete the transactions contemplated by the merger agreement;

its absence of liability for undisclosed broker's fees;

its knowledge of any breach by us of our representations or warranties or of any reason why the conditions to closing the merger would not be timely satisfied; and

its acknowledgment of limitations on our liability for information provided by us to it. **Representations and Warranties about Acquisition Corp.**

Armor Holdings and Acquisition Corp. make various representations and warranties in the merger agreement with respect to Acquisition Corp. that are subject, in some cases, to disclosures and specified exceptions and qualifications. Their joint and several representations and warranties relate to, among other things:

the due organization, valid existence and good standing of Acquisition Corp.;

the capital structure of Acquisition Corp.;

the corporate power and authority of Acquisition Corp. to own, lease and operate its respective properties and carry on its business;

the corporate power and authority of Acquisition Corp. to enter into, execute and deliver the merger agreement and consummate the transactions contemplated thereby;

Acquisition Corp.'s due authorization, execution and delivery of the merger agreement; and

the ability of Acquisition Corp. to satisfy its obligation to pay the merger consideration.

All of the representations and warranties contained in the merger agreement will expire at the effective time of the merger.

Standard for Breach of Representations and Warranties

No representation or warranty made by us (other than our representations and warranties relating to our capitalization and outstanding securities, which shall be true in all respects except for minor variations, and our ordinary course of business liabilities incurred since January 31, 2005, which shall be true in all respects) or Armor Holdings will be deemed untrue, and neither us nor Armor Holdings will be deemed to have breached a representation or warranty, where such failure to be true or breach of such representation or warranty would not be material and adverse to us and our subsidiaries taken as a whole, or Armor Holdings and its subsidiaries taken as a whole, as the case may be.

Covenants Relating to the Conduct of the Business Pending the Merger

From the date of the merger agreement to the effective time of the merger, and unless otherwise provided in the merger agreement or consented to in writing by Armor Holdings, we have agreed to (and will cause our subsidiaries to) carry on our business in the usual and ordinary course in substantially the same manner as prior to signing of the merger agreement, and to use our reasonable best efforts to preserve our business organizations and goodwill, keep available the services of our officers and employees, and maintain satisfactory relationships with those persons with whom we have material and advantageous relationships with.

From the date of the merger agreement to the effective time of the merger, we have agreed, with limited exceptions, that we will not (and we will cause our subsidiaries not to) do any of the following, except as expressly contemplated by the merger agreement or otherwise consented to in writing by Armor Holdings:

enter into any new line of business or change operating policies in any material respect;

pursue or take any new business relating to our Distributed Energy Solutions operations, except as reasonably necessary to fulfill existing contractual obligations;

other than pursuant to outstanding options as of February 27, 2006, issue, deliver, sell, pledge, dispose of or otherwise encumber any of our capital stock or any securities convertible into, or any rights, warrants or options to acquire, any such capital stock;

permit additional shares of our common stock to become subject of new grants under one of our stock option plans or otherwise;

make, declare, set aside or pay dividends on, or make any other distributions in respect of, any of our capital stock, other than our regular quarterly dividend and regular intercompany dividends; split, combine, redeem or reclassify any of our capital stock; or redeem, purchase or otherwise acquire any shares of our capital stock;

sell, transfer, mortgage, encumber or otherwise dispose of any of our assets, business or properties, except for inventories or immaterial obsolete assets or properties in the ordinary course of business;

acquire all or any portion of the assets, business, properties or shares of stock or other securities of any person or entity, other than in the ordinary course;

amend our articles of incorporation, bylaws or any other organizational documents;

implement or adopt any change in financial accounting principles, practices or methods, except as required by generally accepted accounting principles in the United States or other applicable regulatory requirements;

enter into, amend, modify or renew any employment, consulting, change in control or similar agreement, or grant any salary or wage increase, equity awards or incentive or bonus payments, except (1) as required by law or an existing benefit arrangement or (2) for ordinary course merit based or annual salary increases for existing employees or employment arrangements or compensation awards to newly hired or promoted employees, in each case consistent with past practice;

enter into, establish, adopt, amend, modify or renew any employee benefit arrangement or take any action to accelerate the vesting or exercisability of any compensation or benefits payable thereunder, except (1) as required by law or the existing benefit arrangement, (2) amendments that do not increase benefits or result in increased administrative costs or (3) employment arrangements or compensation awards to newly hired or promoted employees consistent with past practice;

settle or compromise any pending or threatened suit, action or claim which would require us or one of our subsidiaries to make a payment of more than \$100,000 or would impose a restriction on the business, assets or operations of the surviving corporation, except that we may make payments relating to matters disclosed in the disclosure schedule provided to Armor Holdings in an amount that has been specifically reserved or accrued therefor in our consolidated balance sheet dated as of January 31, 2006, and if we obtain a general release in connection therewith;

incur, assume, guarantee or otherwise become liable for any debt for borrowed money, or make any loans or advances or make any capital contributions to, or investments in, any other person in excess of \$250,000 individually, or \$500,000 in the aggregate, except for ordinary course trade debt, and for the avoidance of doubt, ordinary course performance bonds, letters of credit and similar instruments;

enter into, modify, amend, terminate (other than at the end of the term) or waive any material rights under any material contract, other than in the ordinary course of business consistent with past practice;

adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of us or any of our subsidiaries (other than the merger);

take, or fail to take, any action which would reasonably be expected to materially delay or prevent clearing any applicable waiting period under the HSR Act;

waive any of our rights in any material respect under any of the confidentiality agreements entered into with any person participating in our process to sell our company; and

agree or commit to do, directly or indirectly, by contract or otherwise, any of the actions listed above.

Reasonable Best Efforts

Each of us, Armor Holdings and Acquisition Corp. have agreed, subject to the terms and conditions in the merger agreement, to use its reasonable best efforts to

take all action and to do all things necessary, proper or advisable to consummate and effect the merger as promptly as practicable and the transactions contemplated by the merger agreement; and

resolve any (1) objections asserted with respect to the transactions under antitrust law or (2) any suits challenging the transactions or otherwise brought under antitrust law or other applicable law that would otherwise materially prohibit or materially impair or materially delay the consummation of the transactions, provided that Armor Holdings is not required to sell or dispose of any business or assets, to the extent that such sale or disposition would have a material adverse effect on Armor Holdings and us, taken as a whole.

Shareholder Approval

As promptly as practicable following the date of the merger agreement, the Company agreed, through its Board of Directors, to duly call, give notice of, convene and hold a special meeting of its shareholders to consider and approve the merger agreement and the merger. At the shareholders' meeting, Parent and Acquisition Corp. will vote all shares owned by them in favor of the merger agreement.

Proxy Statement

The merger agreement provides that we will prepare and work with the SEC to clear this proxy statement as promptly as practicable after the date of the merger agreement. Any filing of, amendment to, or written correspondence with the SEC with respect to this proxy statement is subject to Armor Holdings' review and comment. We agreed to provide Armor Holdings with written copies of any comments by the SEC or its staff to this proxy statement promptly after receipt and copies or notification of our or our counsel's responses to the SEC or its staff. Each of us, Armor Holdings and Acquisition Corp. will use its reasonable best efforts, after consultation with the other parties, to respond promptly to comments of or requests from the SEC and to mail, at the earliest practicable time, the proxy statement and all amendments and supplements thereto to shareholders entitled to vote at the special meeting. We agreed to use our reasonable best efforts to solicit from the shareholders at the special meeting the vote required to approve the merger agreement and the merger and the transactions contemplated thereby.

Press Releases

We and Armor Holdings have agreed to consult with and obtain the consent of the other party before issuing any press release with respect to the merger or the merger agreement, except that such consent is not necessary if the press release or public statement is required by applicable law or securities exchange rules.

Access to Information

From the date of the merger to the effective time, subject to applicable law, we will (and will cause our subsidiaries to) allow Armor Holdings, its representatives and its accountants reasonable access to the books, records and properties and to such other information as Armor Holdings may reasonably request, provided that such access does not unreasonably disrupt our operations.

No Solicitation

We have agreed that we will not, directly or indirectly, through any officer, director, employee, representative or agent of ours or our subsidiaries, and will use commercially reasonable efforts to cause our and our subsidiaries' representatives not to, directly or indirectly:

initiate, solicit or knowingly encourage or facilitate inquiries or proposals with respect to,

participate in negotiations concerning,

provide any confidential information or data to, or

have any substantive discussions with,

any third party relating to, or that would reasonably be expected to lead to, any "acquisition proposal" or enter into any agreement, letter of intent or similar document relating to any acquisition proposal.

The merger agreement defines "acquisition proposal" to mean any proposal or offer (whether or not in writing) with respect to

any purchase of an equity interest (including by means of a tender or exchange offer) representing more than 15% of the voting power in us or any of our subsidiaries,

a merger, consolidation, other business combination, reorganization, recapitalization, dissolution, liquidation or similar transaction involving us or any of our subsidiaries, or

any purchase of assets, businesses, securities or ownership interests (including the securities of any of our subsidiaries) representing more than 15% of our and our subsidiaries' consolidated assets,

in each case other than the transactions contemplated by the merger agreement. We are obligated to notify Armor Holdings promptly of any such acquisition proposal we receive.

We may, however, furnish information to, and participate in discussions or negotiations with, any person that has made a bona fide written acquisition proposal if and only to the extent that:

we first obtain a confidentiality agreement from that person on substantially the same confidentiality, standstill and employee non-solicitation terms with respect to employees as our confidentiality agreement with Armor Holdings;

such acquisition proposal is at a price higher than \$35.00 per share and our Board of Directors believes in good faith that there is a substantial likelihood that the consummation of such acquisition proposal is probable;

after advice of outside legal counsel, our Board of Directors determines in good faith that the failure to take such action would be inconsistent with its fiduciary duties under applicable law; and

we have not violated the no-solicitation provisions of the merger agreement.

If any bona fide written acquisition proposal has not been withdrawn, our Board of Directors may approve or recommend, and we may enter into an agreement providing for, an acquisition proposal that is submitted to us prior to the special meeting, and our Board of Directors may withdraw or modify its recommendation to our shareholders in a manner adverse to Armor Holdings, if:

we have not violated the no-solicitation provisions of the merger agreement;

we provide notice to Armor Holdings at least three days prior to any meeting of our Board of Directors where such action will take place, during which three days Armor Holdings may propose revisions to the terms of the merger agreement;

taking into account Armor Holdings' proposed revisions to the merger agreement, our Board of Directors determines that the acquisition proposal is a "superior proposal"; and

after advice of counsel, our Board of Directors determines in good faith that the failure to take such action would be inconsistent with its fiduciary duties under applicable law.

As used in the merger agreement, the term "superior proposal" means an acquisition proposal made by any person on terms that our Board of Directors determines in good faith, after the receipt of advice from our financial and legal advisors, and considering such factors as our Board of Directors considers to be appropriate (including the conditionality and the timing and likelihood of success of such proposal), are more favorable to us and our shareholders than the transactions contemplated by the merger agreement (including any changes to the terms of the merger agreement proposed by Armor Holdings pursuant to the terms of the merger agreement).

We have agreed to promptly notify Armor Holdings of the receipt of, or any material modifications or material amendments to, any acquisition proposal, or any request (including the terms and conditions of such request) for non-public information relating to us by any person that has notified us of its intention to make an acquisition proposal.

We agreed to immediately terminate any existing discussions or negotiations relating to the foregoing with any persons, other than Armor Holdings, conducted prior to the date of the merger agreement. Additionally, we agreed to immediately request each person who executed a confidentiality agreement in connection with its consideration of acquiring our company return or destroy, in accordance with the terms of such confidentiality agreement, all confidential information furnished to such person by us or on our behalf.

The terms of the merger agreement permit our Board of Directors to withdraw or modify its recommendation to our shareholders in a manner adverse to Armor Holdings if our Board of Directors, determines in good faith, after consulting counsel, that the failure to do so would be inconsistent with its fiduciary duties under applicable law.

Takeover Laws and Provisions

Each party has agreed that it will not take any action that will cause the transactions contemplated by the merger agreement to become subject to any takeover laws and will take all necessary steps within its control to exempt such transactions, or if necessary challenge the validity or applicability of, any applicable takeover law. Each party has agreed that it will not take any action that will cause the transactions contemplated by the merger agreement not to comply with the takeover provisions in the applicable organizational documents of us or our subsidiaries. We also agreed not to elect to become subject to the Texas Business Organizations Code.

Regulatory Applications

We, Armor Holdings and our respective subsidiaries have agreed to:

cooperate with one another to prepare as promptly as possible all documentation, to effect all filings and to obtain all authorizations of all third parties and governmental authorities necessary to consummate the transactions contemplated by the merger agreement as promptly as practicable and make all necessary filings as soon as possible;

upon request, furnish the other party with all information concerning itself, its subsidiaries, directors, officers and shareholders and such other matters as reasonably necessary or advisable in connection with such filing, notice or application made on behalf of such other party or its subsidiaries with or to any third party or governmental authority in connection with such transactions; and

to the extent permitted, give the other party the opportunity to attend and participate in any meetings and conferences with any governmental authority or person objecting to the transactions.

Indemnification

The merger agreement provides that the indemnification provisions in the articles of incorporation and bylaws of the surviving corporation (which shall be the same provisions as exist in our restated articles of incorporation and amended and restated bylaws, as amended, immediately prior to the completion of the merger) shall not be amended, repealed or otherwise modified for six years after the completion of the merger in any manner that would adversely affect the rights thereunder of the individuals who were directors, officers and employees at the effective time of the merger, unless counsel advises that such modification is legally required.

In the merger agreement, Armor Holdings and Acquisition Corp. have agreed to indemnify, defend and hold harmless our and our subsidiaries' current and former directors, officers and employees against all liability arising out of actions or omissions before or at the effective time of the merger to the fullest extent permitted by law and our restated articles of incorporation and amended and restated bylaws, as amended.

Under the merger agreement, Armor Holdings has further agreed that, for a period of six years after the effective time of the merger, it will cause to be maintained in effect director's and officer's liability insurance to reimburse the current and former officers and directors with respect to claims against them arising from facts or events occurring before or at the effective time of the merger. Armor Holdings is obligated to ensure such insurance contains at least the same coverage and amounts of coverage and will contain terms and conditions no less advantageous to the indemnified individuals then those provided in the director's and officer's liability insurance currently provided by us; however, Armor Holdings is not obligated to pay a premium for such coverage in excess of 200% of the current premium, in which event the obligation shall be only to provide the maximum coverage available for a premium not in excess of such amount. The merger agreement further provides that the foregoing obligations shall be satisfied by our purchase on or before the effective time of the merger of a tail-coverage or run-off insurance policy providing the required coverage for the required term and with a premium that does not exceed 200% of the current premium.

If Armor Holdings or any of its successors or assigns: (1) consolidates with or merges into any other entity and is not the continuing or surviving entity or (2) transfers all or substantially all of its assets to any entity, the surviving or transferee entity shall assume the indemnification and insurance obligations discussed above.

Employee Matters

From and after the effective time of the merger until the date six month after the closing of the transactions contemplated by the merger agreement, Armor Holdings has agreed to provide our employees and former employees (including executive officers) taken as a whole with employee benefits and compensation plans, programs and arrangements (including base salary and annual bonus opportunities, but specifically excluding equity grants and any supplemental executive retirement plan benefits) no less favorable, in the aggregate, than those provided by us and our subsidiaries to such persons immediately prior to the effective time of the merger. After such six month period, Armor Holdings has agreed to provide our employees and former employees (including executive officers) taken as a whole with employee benefits and compensation plans, programs and arrangements and severance benefits that are no less favorable, in the aggregate, to those provided to similarly situated employees of Armor Holdings and its subsidiaries. In addition, during the 24 months after the effective time of the merger, any such person whose employment is terminated "without cause" (as defined in such policy) shall be entitled to receive severance payments and benefits under our severance policy disclosed to Armor Holdings.

We and Armor Holdings have also agreed that, subject to Armor Holdings' right to propose modifications to conform to the terms and conditions of its bonus plans for similarly situated



employees, each of our executive officers will be eligible to participate in a bonus plan for the fiscal year ending January 31, 2007 based upon targets and goals substantially similar to those established for our fiscal year ending January 31, 2006, as adjusted for the fiscal year ending January 31, 2007. The proposed terms of the bonus plan provide for a payment to the executive officers (other than Mr. Lukens) if their employment is terminated prior to October 31, 2006 and the year-to-date performance metric through the most recently completed month meets or exceeds the year-to-date forecasted results computed from the quarterly performance metric targets provided by us to Armor Holdings. Armor Holdings has agreed to pay any bonuses earned for the fiscal year ending January 31, 2006, to the extent they we have not paid them as of the effective time.

Financing

We confirmed and agreed in the merger agreement that Wachovia Bank, National Association, and its representatives may provide the financing necessary for Armor Holdings and Acquisition Corp. to pay the merger consideration, although the receipt of such financing is not a condition to Armor Holdings' obligation to consummate the merger.

Conditions to the Merger

The merger agreement contains important conditions to each party's obligation to consummate the merger, including the following:

the approval of the merger agreement and the merger by the holders of at least two-thirds of the outstanding shares of our common stock entitled to vote;

the expiration or termination of any waiting period applicable to the merger under the HSR Act;

the absence of any effective law, rule, regulation, injunction, order, decree or ruling having been enacted, issued, promulgated, enforced or entered by a governmental authority which has the effect of making the merger illegal or otherwise restricting, preventing or prohibiting consummation of the merger;

the other party's representations and warranties in the merger agreement must be true and correct both as of the date of the merger agreement and at and as of the closing date of the merger, as if they were made at and as of the closing date of the merger (except to the extent expressly made as of an earlier date, in which case as of that date) without regard to the standard described in " Representations and Warranties Standard for Breach of Representations and Warranties," except where the aggregate effect of all such failures to be true and correct would not have a "material adverse effect" (as defined below) on the other party; and

the other party's performance in all material respects of all obligations required to be performed by the other party under the merger agreement at or before the effective time of the merger.

In addition, the merger agreement contains important conditions to Armor Holdings' obligation to consummate the merger, including the following:

any approval or consent of any governmental authority necessary for the transactions contemplated by the merger agreement to be consummated, or any relevant governmental waiting periods has been obtained or be effective, except as would not be material to us and our subsidiaries taken as a whole;

no event, change, effect, condition, fact or circumstance has occurred after the date of the merger agreement that, individually or in the aggregate, constitutes or would constitute a "material adverse change" (as defined below); and

there has not been instituted or is not pending any action, proceeding, application, claim or counterclaim by any U.S. federal governmental authority seeking to restrain or prohibit the consummation of the merger.

Definition of "Material Adverse Change" and "Material Adverse Effect"

As used in the merger agreement, a "material adverse effect" or "material adverse change" with respect to us means any effect, change or development that individually, or together with other effects, changes or developments, is (or would be) material and adverse to:

our and our subsidiaries' financial condition, results of operations, assets, liabilities or business, taken as a whole, or

our ability to consummate the transactions contemplated by the merger agreement, excluding in each case the impact of:

changes in laws of general applicability or changes in laws of specific applicability to businesses engaged in contracting with governmental authorities as to the supply of products and services for use in, or related to, military activities ("Defense Businesses") and not disproportionally and materially impacting us, or changes in the interpretation thereof by governmental authorities,

changes in general economic, financial market or political conditions or changes in economic conditions affecting Defense Businesses (including any outbreak or escalation of hostilities or war or any act of terrorism), or

changes resulting from the announcement or the existence of, or compliance with, the merger agreement and the transactions contemplated thereby.

As used in the merger agreement, a "material adverse effect" or "material adverse change" with respect to Armor Holdings or Acquisition Corp. means any effect that would materially impair the ability of Armor Holdings or Acquisition Corp. to perform its obligations under the merger agreement or to consummate the transactions contemplated by the merger agreement by the close of business on November 27, 2006.

Termination

The merger agreement may be terminated and the merger may be abandoned at any time prior to the effective time of the merger as follows:

by mutual written consent of us and Armor Holdings, even if our shareholders have approved and adopted the merger agreement;

by us or Armor Holdings if any governmental authority has enacted, issued, enforced or entered any final and nonappealable injunction, order, decree or ruling which has the effect of making the consummation of the merger illegal or otherwise prevents or prohibits the consummation of the merger; provided that the party seeking to terminate the merger agreement has used its reasonable best efforts to remove or lift such injunction, order, decree or ruling;

by us or Armor Holdings if the merger is not consummated by the close of business on November 27, 2006, provided that such termination right will not be available to any party whose failure to comply with any provision of the merger agreement has resulted in the failure of any conditions to consummating the merger set forth in the merger agreement;

by us if:

Armor Holdings breaches any covenant, agreement, representation or warranty in the merger agreement, if such breach:

would cause any of the conditions described in the fourth and fifth bullets under " Conditions to the Merger" not to be satisfied; and

is incapable of being cured or has not been cured within the 20 days after Armor Holdings receives notice of such breach from us; or

our Board of Directors approves or recommends, or we enter into a definitive agreement, for an acquisition proposal, or our Board of Directors changes its recommendation with respect to the merger agreement, in each case to the extent we have complied with the no-solicitation terms of the merger agreement; and

by Armor Holdings if:

we breach any covenant, agreement, representation or warranty in the merger agreement, if such breach:

would cause any of the conditions described in the fourth and fifth bullets under " Conditions to the Merger" not to be satisfied; and

is incapable of being cured or has not been cured within the 20 days after we receive notice of such breach from Armor Holdings;

we materially violate the no-solicitation terms of the merger agreement to the detriment of Armor Holdings;

our Board of Directors recommends to our shareholders an acquisition proposal or a superior proposal;

we enter into a definitive agreement providing for an acquisition proposal or a superior proposal; or

our Board of Directors:

withdraws the approval of the merger or the merger agreement,

withdraws or modifies its recommendation to the shareholders in a manner adverse to Armor Holdings, or

approves or recommends an acquisition proposal.

Termination Fee

We have agreed to pay Armor Holdings a fee of \$37.34 million in cash if the merger agreement is terminated:

by us if our Board of Directors approves or recommends, or we enter into a definitive agreement for, an acquisition proposal, or our Board of Directors withdraws or modifies its recommendation to our shareholders with respect to the merger agreement in a manner adverse to Armor Holdings; or

by Armor Holdings if:

we materially violate the no-solicitation terms of the merger agreement to the material detriment of Armor Holdings;

our Board of Directors recommends to our shareholders an acquisition proposal or a superior proposal;

we enter into a definitive agreement providing for an acquisition proposal or a superior proposal;

our Board of Directors:

withdraws the approval of the merger or the merger agreement,

withdraws or modifies its recommendation to our shareholders with respect to the merger agreement in a manner adverse to Armor Holdings, or

approves and/or recommends an acquisition proposal; or

we breach any covenant, agreement, representation or warranty in the merger agreement, if:

such breach would cause any of the conditions described in the fourth and fifth bullets under " Conditions to the Merger" not to be satisfied;

such breach is incapable of being cured or has not been cured within the 20 days after we receive notice of such breach from Armor Holdings;

prior to the breach resulting in Armor Holdings' right to terminate the merger agreement, an acquisition proposal has been publicly communicated to our shareholders generally and has not been withdrawn; and

prior to the six month anniversary of the termination of the merger agreement:

we enter into a definitive agreement with respect to the acquisition proposal; or

the acquisition proposal is consummated.

We have agreed to pay Armor Holdings the reasonable expenses actually incurred by it and Acquisition Corp. in connection with the negotiation, preparation, execution and performance of the merger agreement if the merger agreement is terminated by us or Armor Holdings if:

the merger has not been consummated by the close of business on November 27, 2006 because the holders of at least two-thirds of the outstanding shares of our common stock entitled to vote have not approved the merger agreement (and the failure is not a result of the non-terminating party's failure to comply with the terms of the merger agreement);

prior to the applicable meeting of our shareholders, an acquisition proposal has been publicly communicated to our shareholders generally and has not been withdrawn; and

prior to the six month anniversary of the termination of the merger agreement:

we enter into a definitive agreement with respect to the acquisition proposal; or

the acquisition proposal is consummated.

Amendment

At any time prior to the effective time of the merger, any provision of the merger agreement may be waived in writing by the party benefited by such provision or amended or modified by written agreement of the parties to the merger agreement, except to the extent such amendment would violate applicable law or require submission or resubmission of the merger agreement to our shareholders.

MARKET PRICE AND DIVIDEND INFORMATION

Our common stock is traded on The New York Stock Exchange under the symbol "SVC." As of April 5, 2006, we had 29,495,398 shares of common stock outstanding, and we had 492 shareholders of record. The number of shareholders does not take into account those shareholders whose certificates are held by broker-dealers or other nominees. The following table sets forth the range of high and low sales prices of our common stock for each fiscal quarterly period as reported by The New York Stock Exchange and the dividends per share declared by us in each quarterly period within the last two fiscal years and our current fiscal quarter.

Fiscal Year Ended January 31, 2005		High		Low	D	ividend
	_		-			
Fiscal Quarter Ended May 1, 2004	\$	17.20	\$	12.46	\$	0.085
Fiscal Quarter Ended July 31, 2004	\$	18.25	\$	14.02	\$	0.085
Fiscal Quarter Ended October 30, 2004	\$	18.14	\$	14.55	\$	0.085
Fiscal Quarter Ended January 31, 2005	\$	20.63	\$	16.57	\$	0.085
Fiscal Year Ended January 31, 2006		High		Low	D	ividend
					-	
Fiscal Quarter Ended April 30, 2005	\$	25.00	\$	19.74	\$	0.085
Fiscal Quarter Ended July 30, 2005	\$	25.24	\$	20.86	\$	0.085
Fiscal Quarter Ended October 29, 2005	\$	26.00	\$	21.08	\$	0.085
Fiscal Quarter Ended January 31, 2006	\$	26.37	\$	20.00	\$	0.085
Fiscal Year Ending January 31, 2007		High		Low	D	ividend
			_			
Fiscal Quarter Ending April 29, 2006 (through April 5, 2006)(1)	\$	36.97	\$	24.40	\$	N/A

(1)

The highest stock price during this quarter occurred after we announced the execution of the merger agreement. On February 24, 2006, the last trading day before the announcement of the execution of the merger agreement, the closing price of our common stock was \$27.10 per share.

On April 5, 2006, the last reported sales price was \$36.90. You should obtain current market price quotations for shares of our common stock in connection with voting your shares.

Declaration and payment of dividends in the future is dependent upon our earnings and liquidity position and limitations under our revolving credit facility and other debt agreements, among other factors. Based on our financial condition at January 31, 2006, the restrictions imposed by our senior notes and revolving credit facility do not currently restrict our ability to declare and pay dividends at historical levels.

Following the merger there will be no further market for our common stock.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

At the close of business on April 5, 2006, the record date for the special meeting, we had outstanding 29,495,398 shares of common stock. The following table lists the beneficial ownership of shares of our common stock by:

all persons and groups known by us to own beneficially more than 5% of the outstanding shares of our common stock;

each director;

each person who held the office of Chief Executive Officer during the last fiscal year and the four additional highest compensated executive officers who were serving as executive officers on January 31, 2006;

two additional persons each of whom would have been one of such four highest compensated executive officers except that he was not serving as an executive officer on January 31, 2006; and

all current directors and executive officers as a group.

None of our directors or officers owned any equity security issued by our subsidiaries other than director's qualifying shares. Information with respect to current and former officers, directors and their families is as of April 5, 2006 and is based on our books and records and information obtained from each individual. All shareholders set forth below have our principal business address (except as indicated). Additionally, unless otherwise indicated, the owner has sole voting and investment power with respect to the shares indicated.

Name and Address of Beneficial Owner	Shares Owned	Shares Subject to Options(1)	Total Shares Beneficially Owned(2)	Percentage of Shares Outstanding as of April 5, 2006(2)
J. & W. Seligman & Co. Incorporated(3)	1,524,873		1,524,873	5.2%
Ramius Capital Group, L.L.C.(4)	1,654,011		1,654,011	5.6%
Max L. Lukens	31,266	581,000	612,266	2.0%
Carl B. King		213,500	213,500	*
Dennis M. Dellinger	1,762	180,000	181,762	*
L. Scott Biar		27,500	27,500	*
Stephen A. Hines	2,620	64,300	66,920	*
Don K. Kyle(5)		100,000	100,000	*
Mark Whitman(5)	100	40,000	40,100	*
Howard Wolf	21,776	54,000	75,776	*
Monroe M. Luther	9,755	26,000	35,755	*
Charles R. Ofner	4,755	26,000	30,755	*
Charles S. Ream	1,285	10,000	11,285	*
Robert S. Sullivan	4,787	29,000	33,787	*
James M. Tidwell	2,565	10,000	12,565	*
All current executive officers and directors as				
a group (11 persons)	80,571	1,221,300	1,301,871	4.2%

^{*}

Less than 1.0% of the outstanding shares.

(1)

Includes shares issuable upon the exercise of one or more outstanding stock options exercisable within sixty days following April 5, 2006.

(2)

Beneficial ownership is determined in accordance with the rules of the SEC. Shares of our common stock issuable by us pursuant to options that may be exercised within 60 days after April 5, 2006, are deemed to be beneficially owned and outstanding for purposes of calculating the

number of shares and the percentage beneficially owned by the applicable person. However, these shares are not deemed to be beneficially owned and outstanding for purposes of computing the percentage beneficially owned by any other person or entity.

(3)

A Schedule 13G/A filed with the SEC on February 13, 2006 shows J. & W. Seligman Co. Incorporated and William C. Morris, each with a principal business office at 100 Park Avenue, New York, NY 10017, as each having shared voting power over 1,147,505 shares, shared dispositive power over 1,524,873 shares and beneficial ownership of 1,524,873 shares of our common stock. The Schedule 13G/A states that William C. Morris is the owner of a majority of the outstanding voting securities of, and may be deemed to beneficially own the shares reported by, J. & W. Seligman & Co. Incorporated.

(4)

Based solely on a Schedule 13D filed jointly by Starboard Value and Opportunity Master Fund Ltd. ("Starboard"), Parche, LLC ("Parche"), RCG Ambrose Master Fund, Ltd. ("RCG Ambrose"), RCG Halifax Fund, Ltd. ("RCG Halifax"), Ramius Master Fund, Ltd. ("Ramius Master"), Admiral Advisors, LLC ("Admiral Advisors"), Ramius Advisors, LLC ("Ramius Advisors"), Ramius Capital Group, L.L.C. ("Ramius Capital"), C4S & Co., L.L.C. ("C4S"), Peter A. Cohen, Morgan B. Stark, Jeffrey M. Solomon, Thomas W. Strauss, Barington Companies Equity Partners, L.P. ("Barington Companies Equity"), Barington Companies Investors, LLC ("Barington Companies Investors"), Barington Companies Offshore Fund, Ltd. ("Barington Companies Offshore"), Barington Investments, L.P. ("Barington Investments"), Barington Companies Advisors, LLC ("Barington Companies Advisors"), Barington Capital Group, L.P. ("Barington Capital"), LNA Capital Corp. ("LNA") and James Mitarotonda with the Securities Exchange Commission on April 4, 2006. According to the report, Starboard has sole voting and dispositive power with respect to 317,977 shares. Parche has sole voting and dispositive power with respect to 175,186 shares. RCG Ambrose has sole voting and dispositive power with respect to 65,371 shares. RCG Halifax has sole voting and dispositive power with respect to 70,399 shares. Ramius Master has sole voting and dispositive power with respect to 256,456 shares. As the investment manager of Starboard and the managing member of Parche, Admiral Advisors has sole voting and dispositive power with respect to 493,163 shares. As the investment advisor of Ramius Master, Ramius Advisors has sole voting and dispositive power of 256,456 shares. As the sole member of Admiral Advisors and Ramius Advisors and as the investment manager of RCG Halifax and RCG Ambrose, Ramius Capital has sole voting and dispositive power of 885,389 shares. As the managing member of Ramius Capital, C4S has sole voting and dispositive power of 885,389 shares. As the managing members of C4S, each of Messrs. Cohen, Stark, Solomon and Strauss share voting and dispositive power with respect to an aggregate of 885,389 shares owned by Starboard, Parche, RCG Ambrose, RCG Halifax and Ramius Master. Messrs. Cohen, Stark, Solomon and Strauss disclaim beneficial ownership of such shares. Barington Companies Equity has sole voting and dispositive power with respect to 348,448 shares. As the general partner of Barington Companies Equity, Barington Companies Investors has sole voting and dispositive power with respect to 348,448 shares. Barington Companies Offshore has sole voting and dispositive power with respect to 207,936 shares. Barington Investments has sole voting and dispositive power with respect to 212,238 shares. As the investment advisor of Barington Companies Offshore and the general partner of Barington Investments, Barington Companies Advisors has sole voting and dispositive power with respect to 420,174 shares. As the majority member of Barington Companies Investors and the managing member of Barington Companies Advisors, Barington Capital has sole voting and dispositive power with respect to 768,622 shares. As the general partner of Barington Capital, LNA has sole voting and dispositive power with respect to 768,622 shares. Mr. Mitarotonda is the sole stockholder and director of LNA and therefore may be deemed the beneficial owner of shares owned by Barington Companies Equity, Barington Companies Offshore and Barington Investments. Mr. Mitarotonda disclaims beneficial ownership of such shares. Mr. Mitarotonda has sole voting and dispositive power with respect to 768,622 shares. The address

51

of the principal office of each of Parche, Admiral Advisors, Ramius Advisors, Ramius Capital, C4S, Mr. Cohen, Mr. Stark, Mr. Strauss and Mr. Solomon is 666 Third Avenue, 26th Floor, New York, New York 10017. The address of the principal office of each of Starboard, RCG Ambrose, RCG Halifax and Ramius Master is c/o Citco Fund Services (Caymen Islands) Limited, Corporate Center, West Bay Road, Grand Cayman, Cayman Islands, British West Indices. The address of the principal business and principal office of each of Barington Companies Equity, Barington Companies Investors, Barington Investments Barington Companies Advisors, Barington Capital and LNA is 888 Seventh Avenue, 17th Floor, New York, New York 10019. The business address of Mr. Mitarotonda is c/o Barington Capital Group, L.P., 888 Seventh Avenue, 17th Floor, New York, New York, New York 10019. The address of the principal business and principal office of Barington Companies Offshore is c/o Bison Financial Services LTD, Bison Court, Road Town, Tortola, British Virgin Islands.

(5)

Messrs. Kyle and Whitman ceased to be executive officers of Stewart & Stevenson in January 2006.

52

CAUTIONARY STATEMENT CONCERNING FORWARD-LOOKING INFORMATION

This proxy statement contains forward-looking statements, including statements that include the words "believes," "expects," "anticipates," "estimates," "intends," "plans," "projects" or similar expressions and statements regarding our prospects. All statements other than statements of historical fact include in this proxy statement are forward-looking statements. These statements are not guarantees of future performance and involve a number of risks, uncertainties, and assumptions and are made pursuant to the Safe Harbor Provisions of the Private Securities Litigation Reform Act of 1995. Many factors, including those discussed more fully elsewhere herein, in our most recent annual report on Form 10-K, and in our quarterly reports on Form 10-Q, as well as others, could cause results to differ materially from those stated.

Specific important factors that could cause actual results, performance, or achievements to differ materially from such forward-looking statements include risks of dependence on government and failure to obtain new government contracts, inherent risks of government contracts, risks of supply interruptions to Tactical Vehicle Systems segment, risks associated with Distributed Energy Solutions segment, risks of fixed-price contracts, risks as to rising steel prices, risks as to cost controls, risks of general economic conditions, risk of competition, risks relating to technology, risks as to terrorist attacks on the U.S. and their impact on the U.S. economy, risks relating to personnel, risks of claims and litigation, risks of product defects, risks as to foreign sales and global trade matters, risks as to acquisitions and restructuring activities, risks as to currency fluctuations, risks as to environmental and safety matters, and credit risks, all as more specifically outlined in our most recent annual report on Form 10-K. In addition, such forward-looking statements could be affected by general industry and market conditions and growth rates, general domestic and international conditions including interest rates, inflation and currency exchange rates and other future factors. Actual outcomes and results may differ materially from what is expressed or forecasted in such forward-looking statements.

FUTURE SHAREHOLDER PROPOSALS

If the merger is consummated, we will not hold an annual meeting of shareholders in 2006 and there will be no public participation in any future meetings of our shareholders. However, if the merger is not consummated, we will hold a 2006 annual meeting of shareholders. In that event, we will inform our shareholders, by press release or other means we determine to be reasonable, of the date by which shareholder proposals must be received by us for inclusion in the proxy materials relating to the annual meeting, which proposals must comply with the rules and regulations of the SEC then in effect. If a shareholder wishes to present a proposal before the 2006 annual meeting, if held, but does not wish to have the proposal considered for inclusion in our proxy statement and proxy card, that shareholder must also give written notice to our corporate secretary at 2707 N. Loop West, P.O. Box 1637, Houston, Texas 77251-1637. If our corporate secretary does not receive the notice within a reasonable time before we begin to print and mail our proxy materials for such meeting, the proxies designated by our Board of Directors will have discretionary authority to vote on any such proposal.

In the event the merger is not consummated and we hold a 2006 annual meeting of shareholders, a shareholder's notice to our corporate secretary shall set forth as to each matter the shareholder proposes to bring before the annual meeting:

a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting,

the name and address, as they appear on our books, of the shareholder proposing such business,

the class and number of shares of our voting stock which are beneficially owned by the shareholder,

a representation that the shareholder intends to appear in person or by proxy at the meeting to bring the proposed business before the annual meeting, and

a description of any material interest of the shareholder in such business.

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and special reports, proxy statements and other information with the SEC. Our SEC filings are available to the public over the Internet at the SEC's web site at http://www.sec.gov. In addition, you may read and copy any reports, statements or other information that we file with the SEC at the SEC's public reference room at the following location:

Public Reference Room 100F Street, Room 2521 Washington, D.C. 20549

Please call the SEC at 1-800-SEC-0330 for further information regarding the public reference room.

Statements contained in this proxy statement regarding the contents of any contract or other document are not necessarily complete and each such statement is qualified in its entirety by reference to that contract or other document filed as an exhibit with the SEC.

You should rely only on the information contained in this proxy statement. We have not authorized anyone to provide you with information that is different from what is contained in this proxy statement. This proxy statement is dated April 6, 2006. You should not assume that the information contained in this proxy statement is accurate as of any date other than that date, and the mailing of the proxy statement to shareholders does not create any implication to the contrary.

ADDITIONAL INFORMATION

To vote your shares, please complete, date, sign and return the enclosed proxy card as soon as possible in the enclosed postage-prepaid envelope. If you would like additional copies of this proxy statement, without charge, or if you have questions about the merger, including the procedures for voting your shares, you should contact:

MacKenzie Partners, Inc. 105 Madison Avenue 14th Floor New York, NY 10016 Call Toll-Free (800) 322-2885 or Call Collect (212) 929-5500 E-mail: proxy@mackenziepartners.com

You may also wish to consult your own legal, tax and/or financial advisors with respect to any aspect of the merger, the merger agreement or other matters discussed in this proxy statement.

If you would like to request additional information from us, please do so by April 24, 2006 in order to receive the information before the special meeting.



APPENDIX A

AGREEMENT AND PLAN OF MERGER

DATED AS OF FEBRUARY 27, 2006

BY AND AMONG

ARMOR HOLDINGS, INC.,

SANTANA ACQUISITION CORP.

AND

STEWART & STEVENSON SERVICES, INC.

TABLE OF CONTENTS

		Page
ARTICLE I	DEFINITIONS; INTERPRETATION	A-4
1.01	Definitions	A-4
1.01	Interpretation	A-10
1.02	merpretation	A-10
ARTICLE II	THE MERGER	A-11
2.01	The Merger	A-11
2.02	Closing	A-11
2.03	Effective Time	A-11
2.04	Effects of the Merger	A-11
2.05	Articles of Incorporation and By-laws	A-11
2.06	Directors and Officers	A-11
ARTICLE III	CONSIDERATION; EXCHANGE PROCEDURES	A-12
3.01	Conversion or Cancellation of Shares	A-12
3.02	Exchange of Certificates; Payment of the Merger Consideration	A-12
3.03	Stock Options	A-14
3.04	Dissenting Shareholders	A-14
ARTICLE IV	CONDUCT OF BUSINESS PENDING THE MERGER	A-15
		. 15
4.01	Agreements of the Company	A-15
4.02	Agreements of Parent	A-17
		A 17
ARTICLE V	REPRESENTATIONS AND WARRANTIES	A-17
5.01	Disclosure Schedules	A-17
5.02	Standard for Breach of Representations and Warranties	A-18
5.03	Representations and Warranties	A-18
5.04	Representations and Warranties about Merger Sub	A-28
0.01		
ARTICLE VI	COVENANTS	A-29
6.01	Reasonable Best Efforts	A-29
6.02	Shareholder Approvals	A-29
6.03	Filings	A-29
6.04	Press Releases	A-30
6.05	Access; Information	A-30
6.06	No Solicitation	A-30
6.07	Takeover Laws and Provisions	A-32
6.08	Regulatory Applications	A-32
6.09	Indemnification	A-33
6.10	Employee Matters	A-35
6.11	Notification of Certain Matters	A-37
6.12	Parent Financing	A-37

ARTICLE VII	CONDITIONS TO THE MERGER	A-37
7.01	Conditions to Each Party's Obligation to Effect the Merger	A-37
7.02	Conditions to the Company's Obligation	A-37
7.03	Conditions to Parent's Obligation	A-38
ARTICLE	TERMINATION	A-39
VIII		
8.01	Termination	A-39
8.02	Effect of Termination	A-40
8.03	Expenses	A-40
	•	
ARTICLE IX	MISCELLANEOUS	A-41
9.01	Survival	A-41
9.02	Waiver; Amendment	A-41
9.03	Counterparts	A-41
9.04	Governing Law; Jurisdiction; Venue	A-41
9.05	Notices	A-41
9.06	Entire Understanding; No Third Party Beneficiaries	A-42
9.07	Severability	A-42
9.08	Assignment; Successors	A-43
Schedules		

Schedule 2.06Officers of the Surviving CorporationSchedule 6.01Contract ConsentsCompany's Disclosure Schedule

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this "*Agreement*"), dated as of February 27, 2006, is by and among Armor Holdings, Inc., a Delaware corporation ("*Parent*"), Santana Acquisition Corp., a Texas corporation ("*Merger Sub*"), and Stewart & Stevenson Services, Inc., a Texas corporation (the "*Company*").

RECITALS

A. *The Proposed Transaction.* The parties intend to effect a strategic business combination through the merger of Merger Sub with and into the Company (the "*Merger*"), with the Company being the surviving corporation (the "*Surviving Corporation*").

B. *Board Determinations.* The respective boards of directors of Parent, Merger Sub and the Company have each determined that the Merger and the other transactions contemplated hereby are in the best interests of their respective stockholders or shareholders, as the case may be, and, therefore, have approved the Merger and this Agreement and the plan of merger contained in this Agreement (the "*Plan of Merger*"), and the respective boards of directors of Merger Sub and the Company have each recommended that this Agreement and the Merger be approved by the respective shareholders of Merger Sub and the Company.

NOW, THEREFORE, in consideration of the premises, and of the mutual representations, warranties, covenants and agreements contained in this Agreement, Parent, Merger Sub and the Company agree as follows:

ARTICLE I DEFINITIONS; INTERPRETATION

1.01 Definitions. This Agreement uses the following definitions:

"Acquisition Proposal" means any proposal or offer (whether or not in writing) with respect to (a) any purchase of an equity interest (including by means of a tender or exchange offer) representing more than 15% of the voting power in the Company or any of its Subsidiaries, (b) a merger, consolidation, other business combination, reorganization, recapitalization, dissolution, liquidation or similar transaction involving the Company or any of its Subsidiaries or (c) any purchase of assets, businesses, securities or ownership interests (including the securities of any Subsidiary of the Company) representing more than 15% of the consolidated assets of the Company and its Subsidiaries, other than the Transactions.

"Agreement" has the meaning assigned in the Preamble to this Agreement.

"Articles of Merger" has the meaning assigned in Section 2.03.

"*Benefit Arrangement*" means, with respect to the Company, each of the following (a) under which any Employee or any of its current or former directors has any present or future right to benefits, (b) that is sponsored or maintained by it or its Subsidiaries, or (c) under which it or its Subsidiaries has had or has any present or future liability: each "employee benefit plan"

(within the meaning of Section 3(3) of ERISA) and each stock purchase, stock option, equity-based grants, severance, employment, change-in-control, fringe benefit, bonus, incentive, deferred compensation, paid time off benefits and other employee benefit plan, agreement, program, policy or other arrangement (with respect to any of the preceding, whether or not subject to ERISA).

"Benefits Transition Date" has the meaning assigned in Section 6.10(a).

"Business Day" means any day other than a day on which banks in the State of Texas are required or authorized to be closed.

"Certificate" has the meaning assigned in Section 3.01(a).

"Certificate of Merger" has the meaning assigned in Section 2.03.

"Closing" has the meaning assigned in Section 2.02.

"Closing Date" has the meaning assigned in Section 2.02.

"*Code*" has the meaning assigned in Section 5.03(m)(2).

"Company" has the meaning assigned in the Preamble to this Agreement.

"Company Board" means the Board of Directors of the Company.

"Company Board Recommendation" has the meaning assigned in Section 5.03(e).

"Company Board Change of Recommendation" has the meaning assigned in Section 6.06(b).

"Company Common Stock" means the common stock, without par value, of the Company.

"Company Representatives" has the meaning assigned in Section 6.06(a).

"Company Shareholder Approval" has the meaning assigned in Section 5.03(e).

"Company Stock Option" has the meaning assigned in Section 3.03.

"Company Stock Plans" has the meaning assigned in Section 3.03.

"Confidentiality Agreement" means that certain Confidentiality Agreement dated as of December 7, 2005 between Parent and the Company.

"*Constituent Documents*" means the charter or articles or certificate of incorporation and by-laws of a corporation, the certificate of partnership and partnership agreement of a general or limited partnership, the certificate of formation and limited liability company agreement of a limited liability company, the trust agreement of a trust and the comparable documents of other entities.

"Covered Employees" has the meaning assigned in Section 6.10(a).

"Defense Businesses" has the meaning assigned in the definition of "Material Adverse Effect" in this Section 1.01.

"Disbursing Agent" has the meaning assigned in Section 3.02(a).

"Disclosure Schedule" has the meaning assigned in Section 5.01.

"Dissenting Shareholder" has the meaning assigned in Section 3.04(a).

"*Dissenting Shares*" means shares of Company Common Stock the holders of which have perfected and not withdrawn or lost their right to dissent with respect to such shares under Articles 5.11, 5.12, and 5.13 and of the TBCA, as applicable.

"Effective Time" has the meaning assigned in Section 2.03.

"Eligible Option" has the meaning assigned in Section 3.03.

"Employees" means current and former employees of the Company and its Subsidiaries.

"*Environmental Laws*" means the statutes, rules, regulations, ordinances, codes, orders, decrees, and any other laws (including common law) of any foreign, federal, state, local and any other governmental authority, regulating, relating to or imposing liability or standards of conduct concerning pollution or protection of human health and safety or of the environment, in every case as in effect on or prior to the date of this Agreement.

"ERISA" means the U.S. Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" has the meaning assigned in Section 5.03(m)(3).

"Exception Shares" means, collectively, shares of Company Common Stock owned or held by the Company, Parent or Merger Sub and/or any of their respective Subsidiaries.

"Exchange Act" means the U.S. Securities Exchange Act of 1934 and the rules and regulations thereunder.

"Existing Confidentiality Agreements" has the meaning assigned in Section 5.03(w).

"Financing" has the meaning assigned in Section 6.12.

"GAAP" means generally accepted accounting principles in the United States.

"Governmental Authority" means any court, administrative agency or commission or other governmental authority or instrumentality, domestic or foreign.

"HSR Act" means the U.S. Hart-Scott-Rodino Antitrust Improvements Act of 1976 and the rules and regulations thereunder.

"Indemnified Party" has the meaning assigned in Section 6.09(b).

"Information Memorandum" has the meaning assigned in Section 5.03(s)(3).

"*Knowledge*" and "*knowingly*" means or has reference to, respectively, the actual knowledge of the executive officers of the Company or Parent, as the case may be, after due inquiry consistent with their respective responsibilities as an executive officer.

"*Lien*" means any charge, mortgage, pledge, security interest, restriction, claim, lien, or encumbrance other than, with respect to the Company or its Subsidiaries, (a) any such charge, mortgage, pledge, security interest, restriction, claim, lien or encumbrance disclosed in the consolidated financial statements of the Company and its Subsidiaries or the notes thereto or securing liabilities reflected on such financial statements or incurred in the ordinary course of business since the date of the most recent consolidated financial statements, (b) any such charge, mortgage, pledge, security interest, restriction, claim, lien or encumbrance for Taxes not yet delinquent or that are being contested in good faith and reserved for in accordance with GAAP, or (c) any such charge, mortgage, pledge, security interest, restriction, claim, lien or encumbrance which is a carrier's, warehousemen's, mechanic's, materialmen's, repairmen's or other similar lien arising in the ordinary course of business.

"Management Presentation" has the meaning assigned in Section 5.03(s)(3).

"Material Adverse Effect" or "Material Adverse Change" means

(a) with respect to the Company, any effect, change or development that individually, or together with other effects, changes or developments, is (or would be) material and adverse to (i) the financial condition, results of operations, assets, liabilities or business of the Company and its Subsidiaries, taken as a whole or (ii) the ability of the Company to consummate the Transactions, excluding in each case the impact of (1) changes in laws of general applicability or changes in laws of specific applicability to businesses engaged in contracting with Governmental Authorities as to the supply of products and services for use in, or related to, military activities ("*Defense Businesses*") and not disproportionally and materially impacting the Company, or changes in the interpretation thereof by Governmental Authorities, (2) changes in general economic, financial market or political conditions or changes in economic conditions affecting Defense Businesses (including any outbreak or escalation of hostilities or war or any act of terrorism), or (3) changes resulting from the announcement or the existence of, or compliance with, this Agreement and the Transactions; and

(b) with respect to Parent or Merger Sub, any effect that would materially impair the ability of Parent or Merger Sub to perform its obligations under this Agreement or to consummate the Transactions by the Termination Date.

"Material Contract" has the meaning assigned in Section 5.03(k)(1).

"*Materials of Environmental Concern*" means any hazardous or toxic substances, materials, wastes, pollutants, or contaminants, including without limitation those defined or regulated as such under any Environmental Law, and any other substance the presence of which may give rise to liability under any Environmental Law.

"Merger" has the meaning assigned in the Recitals.

"Merger Sub" has the meaning assigned in the Preamble to this Agreement.

"Merger Sub By-laws" means the by-laws of Merger Sub.

"Merger Sub Common Stock" means the common stock, \$0.001 par value per share, of Merger Sub.

"Parent" has the meaning assigned in the Preamble to this Agreement.

"Parent Board" means the Board of Directors of Parent.

"party" means Parent, Merger Sub or the Company.

"Pension Plan" has the meaning assigned in Section 5.03(m)(2).

"Per Share Consideration" has the meaning assigned in Section 3.01(a).

"person" is to be interpreted broadly to include any individual, corporation, limited liability company, partnership, association, joint-stock company, business trust or unincorporated organization.

"Plan of Merger" has the meaning assigned in the Recitals.

"*Previously Disclosed*" means information set forth by the Company in the applicable paragraph of the Disclosure Schedule, or any other paragraph of the Disclosure Schedule (so long as it is reasonably clear from the context that the disclosure in such other paragraph of the Disclosure Schedule is also applicable to the Section or subsection of this Agreement in question) or in the Company's Regulatory Filings (including any schedules and exhibits thereto).

"Proxy Statement" has the meaning assigned in Section 5.03(t).

"Regulatory Filings" has the meaning assigned in Section 5.03(g)(1).

"*Representatives*" means, with respect to any person, such person's directors, officers, employees, legal or financial advisors or any representatives of such legal or financial advisors.

"Requisite Approvals" has the meaning assigned in Section 6.08(a).

"*Rights*" means, with respect to any person, securities or obligations convertible into or exercisable or exchangeable for, or giving any other person any right to subscribe for or acquire, or any options, calls or commitments relating to, or any stock appreciation right or other instrument the value of which is determined in whole or in part by reference to the market price or value of, shares of capital stock of such first person.

"Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002 and the rules and regulations thereunder.

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the U.S. Securities Act of 1933 and the rules and regulations promulgated thereunder.

"Shareholders' Meeting" has the meaning assigned in Section 6.02.

"Subsidiary" has the meaning ascribed to such term in Rule 1-02 of Regulation S-X promulgated by the SEC.

"Superior Proposal" means an Acquisition Proposal made by any person on terms that the Company Board determines in good faith, after the receipt of advice from the Company's financial and legal advisors, and considering such factors as the Company Board considers to be appropriate (including the conditionality and the timing and likelihood of success of such proposal), are more favorable to the Company and its shareholders than the Transactions (including any changes to the terms of this Agreement proposed by Parent in response to such offer or otherwise pursuant to Section 6.06(b)).

"Surviving Corporation" has the meaning assigned in the Recitals.

"Takeover Laws" has the meaning assigned in Section 5.03(n).

"Takeover Provisions" has the meaning assigned in Section 5.03(n).

"*Tax*" and "*Taxes*" means all federal, state, local or foreign taxes, charges, fees, levies or other assessments, however denominated, including, without limitation, all net income, gross income, gains, gross receipts, sales, use, ad valorem, goods and services, capital, production, transfer, franchise, windfall profits, license, withholding, payroll, employment, disability, employer health, excise, estimated, severance, stamp, occupation, property, environmental, unemployment or other taxes, custom duties, fees, assessments or charges of any kind whatsoever, together with any interest and any penalties, additions to tax or additional amounts imposed by any taxing authority.

"*Tax Returns*" means any return, amended return or other report (including elections, declarations, disclosures, schedules, estimates and information returns) required to be filed with respect to any Tax.

"TBCA" means the Texas Business Corporation Act.

"TBOC" means the Texas Business Organizations Code.

"Termination Date" has the meaning assigned in Section 8.01(d).

"Texas Courts" has the meaning assigned in Section 9.04.

"Transactions" has the meaning assigned in Section 5.03(e).

1.02 Interpretation.

(a) In this Agreement, except as the context may otherwise require, references:

(1) to the Preamble, Recitals, Sections, Annexes or Schedules are to the Preamble to, a Recital or Section of, or Annex or Schedule to, this Agreement;

(2) to this Agreement are to this Agreement and the Annexes and Schedules to it taken as a whole;

(3) to any agreement (including this Agreement), contract, statute or regulation are to the agreement, contract, statute or regulation as amended, modified, supplemented, restated or replaced from time to time (in the case of an agreement or contract, to the extent permitted by the terms thereof);

- (4) to any section of any statute or regulation include any successor to the section;
- (5) to any Governmental Authority include any successor to that Governmental Authority; and
- (6) to the date of this Agreement are to February 27, 2006.

(b) The table of contents and Article and Section headings are for reference purposes only and do not limit or otherwise affect any of the substance of this Agreement.

(c) The words "include," "includes" or "including" are to be deemed followed by the words "without limitation".

(d) The words "herein", "hereof" or "hereunder", and similar terms are to be deemed to refer to this Agreement as a whole and not to any specific Section.

(e) This Agreement is the product of negotiation by the parties, which have had the assistance of counsel and other advisers. The parties intend that this Agreement not be construed more strictly with regard to one party than with regard to another.

(f) No provision of this Agreement is to be construed to require, directly or indirectly, any person to take any action, or omit to take any action, to the extent such action or omission would violate applicable law (including statutory and common law), rule or regulation.

ARTICLE II THE MERGER

2.01 *The Merger*. Upon the terms and subject to the conditions set forth in this Agreement and in accordance with the provisions of the TBCA and the TBOC, Merger Sub will merge with and into the Company at the Effective Time. At the Effective Time the separate corporate existence of Merger Sub will terminate. The Company will be the Surviving Corporation and will continue its corporate existence under the laws of the State of Texas.

2.02 *Closing*. The closing of the Merger (the "*Closing*") will take place in the offices of Fulbright & Jaworski L.L.P., 1301 McKinney St., Suite 5100, Houston, Texas, at 10:00 a.m. as promptly as practicable (but in any event within two Business Days) after satisfaction or waiver of the conditions set forth in Article VII, other than those conditions that by their nature are to be satisfied at the Closing but subject to the fulfillment or waiver of those conditions (the "*Closing Date*").

2.03 *Effective Time.* Subject to the provisions of this Agreement, at the Closing the Company and Merger Sub will duly execute and file articles of merger or a certificate of merger, as applicable (collectively, the "*Articles of Merger*") with the Secretary of State of the State of Texas for filing under Article 5.04 of the TBCA or Section 10.153 of the TBOC as required. The parties will make all other filings or recordings required under the TBCA or the TBOC, and the Merger will become effective when the Articles of Merger are filed in the office of the Secretary of State of the State of Texas and a certificate of merger (the "*Certificate of Merger*") is issued by the office of the Secretary of State of the State of Texas pursuant to Article 5.05 of the TBCA, or at such later date or time as Parent and the Company shall agree and specify in the Articles of Merger in accordance with Article 10.03 of the TBCA (the time the Merger becomes effective being the "*Effective Time*").

2.04 *Effects of the Merger.* The Merger will have the effects prescribed by the TBCA, the TBOC and other applicable law.

2.05 Articles of Incorporation and By-laws.

(a) The amended and restated articles of incorporation of the Company, as in effect immediately before the Effective Time, will be the articles of incorporation of the Surviving Corporation as of the Effective Time, except that the name of the Surviving Corporation shall be changed therein to be the name of the Company immediately prior to the Effective Time.

(b) The Merger Sub By-laws, as in effect immediately before the Effective Time, will be the by-laws of the Surviving Corporation as of the Effective Time except that the Merger Sub By-laws shall, to the extent that they do not do so, be amended (and be deemed to be so amended) as of the Effective Time to effectuate the obligations of Parent and the Surviving Corporation provided in Sections 6.09(a) and 6.09(b).

2.06 *Directors and Officers.* The directors of Merger Sub immediately prior to the Effective Time shall be the directors of the Surviving Corporation as of the Effective Time. The

officers of the Surviving Corporation as of the Effective Time and their titles, subject to the right of the board of directors of the Surviving Corporation to appoint or replace officers, shall be as set forth in Section 2.06 of the Disclosure Schedule.

ARTICLE III CONSIDERATION; EXCHANGE PROCEDURES

3.01 *Conversion or Cancellation of Shares.* At the Effective Time, by virtue of the Merger and without any action on the part of any shareholder:

(a) Each share of Company Common Stock issued and outstanding immediately prior to the Effective Time, other than Exception Shares (which shares shall be cancelled and shall cease to exist with no payment being made with respect thereto) and Dissenting Shares (which shares shall be treated in accordance with Section 3.04) shall be converted into and constitute the right to receive cash in an amount equal to \$35.00, without interest (the "*Per Share Consideration*"). At the Effective Time, all shares of Company Common Stock that have been converted into the right to receive the Per Share Consideration as provided in this Section 3.01(a) shall no longer be outstanding and shall be cancelled and shall cease to exist, and each holder of a certificate that immediately prior to the Effective Time represented such shares of Company Common Stock (a "*Certificate*") shall cease to have any rights with respect thereto, except the right to receive (i) the Per Share Consideration and (ii) any dividends with respect to Company Common Stock with a record date prior to the Effective Time but unpaid as of the Effective Time.

(b) Each issued and outstanding share of Merger Sub Common Stock shall be converted into one fully paid and nonassessable share of common stock, without par value, of the Surviving Corporation.

3.02 Exchange of Certificates; Payment of the Merger Consideration.

(a) Appointment of Disbursing Agent. At or prior to the Effective Time, Parent shall deposit with a disbursing agent agreed upon by Parent and the Company (the "*Disbursing Agent*") cash in an amount sufficient to allow the Disbursing Agent to make all payments that may be required pursuant to this Article III. Parent shall be obligated to, from time to time, deposit any additional funds necessary to make all payments that may be required by this Article III. Such funds shall be invested by the Disbursing Agent in short term investments in direct obligations of the United States of America, obligations for which the full faith and credit of the United States of America is pledged to provide for the payment of all principal and interest or commercial paper obligations receiving the highest rating from either Moody's Investors Service, Inc. or Standard & Poor's or a combination thereof as directed by Parent; provided that no such investment or loss thereon shall affect the amounts payable or the timing of the amounts payable pursuant to this Article III. Any net profits resulting from, or interest or income produced by, such investments shall be payable as directed by the Parent. Upon the first anniversary of the Effective Time, any such cash remaining in the possession of the

Disbursing Agent (together with any earnings in respect thereof) shall be delivered to Parent and any holder of Certificates who has not theretofore exchanged such Certificates pursuant to this Article III shall thereafter be entitled to look exclusively to Parent and/or the Surviving Corporation, and only as a general creditor thereof, for the consideration to which such holder may be entitled upon exchange of such Certificates pursuant to this Article III. Notwithstanding the foregoing, neither the Disbursing Agent nor any party hereto shall be liable to any holder of Certificates for any amount properly delivered to a public official pursuant to applicable abandoned property, escheat or similar laws.

(b) Exchange Procedures. Promptly after the Effective Time, but in no event later than two Business Days thereafter, Parent shall cause the Disbursing Agent to mail or deliver to each person who was, immediately prior to the Effective Time, a holder of record of Company Common Stock, a form of letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to Certificates shall pass, only upon proper delivery of such certificates to the Disbursing Agent) containing instructions for use in effecting the surrender of Certificates in exchange for the consideration to which such person is entitled pursuant to this Article III. Upon surrender to the Disbursing Agent of a Certificate for cancellation together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, the holder of such Certificate shall promptly be provided in exchange therefor cash in the amount to which such holder is entitled pursuant to this Article III, and the Certificate so surrendered shall forthwith be canceled. No interest will accrue or be paid with respect to any consideration to be delivered upon surrender of Certificates.

(c) Transfer to Holder other than Existing Holder. If any cash payment is to be made in a name other than that in which the Certificate surrendered in exchange therefor is registered, it shall be a condition of such payment that the person requesting such payment shall pay any transfer or other similar Taxes required by reason of the making of such payment in a name other than that of the registered holder of the Certificate surrendered, or required for any other reason relating to such holder or requesting person, or shall establish to the reasonable satisfaction of the Disbursing Agent that such Tax has been paid or is not payable.

(d) Transfers. At or after the Effective Time, there shall be no transfers registered on the stock transfer books of the Surviving Corporation of Company Common Stock or Certificates that were outstanding immediately prior to the Effective Time.

(e) Lost, Stolen or Destroyed Certificates. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation or the Disbursing Agent, the posting by such person of a bond in such reasonable amount as the Surviving Corporation or the Disbursing Agent may direct as indemnity against any claim that may be made against it with respect to such Certificate, the Surviving Corporation or the Disbursing Agent shall, in exchange for such lost, stolen or destroyed Certificate, pay or cause to be paid the consideration deliverable in respect of Company Common Stock formerly represented by such Certificate pursuant to this Article III.

3.03 *Stock Options.* The Company shall (a) terminate the Company's Non-statutory Stock Option Plan (as amended and restated effective as of June 10, 1997, and as subsequently amended), Amended and Restated 1993 Non-officer Employee Stock Option Plan, 1994 Director Stock Option Plan, and Amended and Restated 1996 Director Stock Plan (as amended by Amendment No. 1 effective as of July 9, 2004) (collectively, the "*Company Stock Plans*"), immediately prior to the Effective Time without prejudice to the rights of the holders of options (each, a "*Company Stock Option*") awarded pursuant thereto, (b) following such termination grant no additional options under the Company Stock Plans or permit the receipt of shares of Company Common Stock, and (c) cause each Company Stock Option that is outstanding immediately prior to the consummation of the Merger to become fully vested and exercisable. Prior to the Effective Time, the Company and Parent will take all actions reasonably necessary to provide that, upon the Effective Time, each outstanding Company Stock Option shall be cancelled automatically and at the Effective Time shall be converted into and constitute the right to receive cash in an amount equal (less any applicable withholding and without interest) to the product of (1) the total number of shares of Company Common Stock subject to such holder's Company Stock Option or Options immediately prior to the Effective Time and (2) the excess, if any, of the Per Share Consideration over the exercise price per share of Company Common Stock subject to such Company Stock Option or Options (each, an "*Eligible Option*"). No payment of the Per Share Consideration with respect to an Eligible Option shall be made by the Disbursing Agent to the holder of such Eligible Option until receipt by the Disbursing Agent of an option cancellation agreement, in a form mutually and reasonably acceptable to the Company and Parent, with respect to all Eligible Options owned by the holder of such Eligible Option.

3.04 Dissenting Shareholders.

(a) Each Dissenting Share shall not be converted into or represent a right to receive the Per Share Consideration hereunder, and the holder thereof shall be entitled only to such rights as are granted by Articles 5.11, 5.12, and 5.13 of the TBCA, as applicable. The Company shall give Parent notice as promptly as reasonably practicable upon receipt by the Company of any demand for payment pursuant to Article 5.12 of the TBCA, as applicable, and of withdrawals of such notice (any shareholder duly making such demand being hereinafter called a "*Dissenting Shareholder*"), and Parent shall have the right to participate in and direct all negotiations and proceedings with respect to any such demands. The Company shall not, without the prior written consent (which consent may not be unreasonably withheld or delayed) of Parent, make any payment with respect to, settle, offer to settle or otherwise negotiate, any such demands. Any payments made in respect of Dissenting Shares shall be made by Parent and/or the Surviving Corporation.

(b) If any Dissenting Shareholder shall have withdrawn or lost (through failure to perfect or otherwise) his or her right to dissent under Articles 5.11, 5.12, and 5.13 of the TBCA, as applicable, each of such holder's shares of Company Common Stock shall be converted solely into a right to receive the Per Share Consideration, without interest, in accordance with the applicable provisions of this Agreement.

ARTICLE IV CONDUCT OF BUSINESS PENDING THE MERGER

4.01 *Agreements of the Company.* The Company agrees that from the date of this Agreement until the Effective Time, except as expressly contemplated by this Agreement or as Previously Disclosed, without the prior written consent of Parent (which consent will not be unreasonably withheld or delayed), it will not, and will cause each of its Subsidiaries not to:

(a) Ordinary Course. Conduct its business and the business of its Subsidiaries other than in the ordinary and usual course consistent with past practices in all material respects and in compliance in all material respects with applicable laws, or fail to use commercially reasonable efforts to (i) preserve intact, in all material respects, their business organizations and goodwill, (ii) keep available the services of its officers and employees, and (iii) maintain satisfactory relationships with those persons having material and advantageous business relationships with the Company or its Subsidiaries.

(b) Operations. Enter into any new line of business or change its operating policies in any material respect except as required by applicable law, regulation or policies imposed by any Governmental Authority; provided, however, that the Company will not be deemed to be engaged in a new line of business to the extent it prosecutes its efforts to sell trucks, trailers and related kinds of products and services reasonably related to its current line of business. The Company will not pursue or take any new business related to its Distributed Energy Solutions operations, except to the extent reasonably necessary to fulfill its existing contractual obligations.

(c) Capital Stock and Other Securities. Other than pursuant to Rights Previously Disclosed and outstanding on the date of this Agreement, (1) issue, sell or otherwise permit to become outstanding or dispose of or encumber or pledge, or authorize or propose the creation of, any additional shares of its stock or any other securities (including long-term debt) or any Rights with respect to its stock or any other securities, or (2) permit any additional shares of its stock to become subject to new grants under Company Stock Plans or otherwise.

(d) Dividends, Distributions, Repurchases. (1) Make, declare, pay or set aside for payment any dividend on or in respect of, or declare or make any distribution on any shares of its stock, other than (A) dividends from its wholly owned Subsidiaries to it or another of its wholly owned Subsidiaries consistent with past practice or (B) regular quarterly dividends on Company Common Stock in an amount not exceeding \$.085 per share and consistent with past practices with respect to the timing of the payment of such dividends, or (2) directly or indirectly adjust, split, combine, redeem or reclassify, or purchase or otherwise acquire any shares of its stock.

(e) Dispositions. Sell, transfer, mortgage, encumber or otherwise dispose of any of its assets, business or properties, except for sales, transfers, mortgages, encumbrances or other dispositions of inventories or de minimis obsolete assets or properties in the ordinary course of business.

(f) Acquisitions. Acquire all or any portion of the assets, business, properties or shares of stock or other securities of any other person other than the purchase of assets and properties in the ordinary course of business.

(g) Constituent Documents. Amend its Constituent Documents.

(h) Accounting Methods. Implement or adopt any change in its financial accounting principles, practices or methods, other than as may be required by GAAP or regulatory accounting requirements applicable to U.S.-publicly owned business organizations generally.

(i) Compensation; Employment Agreements; Etc. Enter into, amend, modify or renew any employment, consulting, change in control or similar contract, agreement or arrangement with any director or employee, or grant any salary or wage increase, equity awards or incentive or bonus payments, except (1) to make changes that are required by applicable law or the terms of a Benefit Arrangement, (2) to grant merit-based or annual salary increases in the ordinary and usual course of business and in accordance with past practice or (3) for employment arrangements for, or grants of awards to, newly hired employees or promoted employees in the ordinary and usual course of business consistent with past practice.

(j) Benefit Plans. Enter into, establish, adopt, amend, modify or renew any pension, retirement, stock option, stock purchase, savings, profit sharing, deferred compensation, bonus, group insurance or other employee benefit, incentive or welfare contract, plan or arrangement or any trust agreement in respect of any director, officer or employee or take any action to accelerate the vesting or exercisability of stock options, restricted stock or other compensation or benefits payable thereunder, except (1) as may be required by applicable law or the terms of a Benefit Arrangement, (2) amendments that do not increase benefits or result in increased administrative costs or (3) for employment arrangements for, or grants of awards to, newly hired employees or promoted employees in the ordinary and usual course of business consistent with past practice.

(k) Litigation. Settle or compromise any litigation, or other pending or threatened suit, action, or claim which would require the payment by the Company or a Subsidiary of more than \$100,000 or would impose a restriction on the business, assets or operations of the Company following the Effective Time, except that the Company may make any such payment in respect of matters Previously Disclosed in an amount in each such case to the extent specifically reserved or accrued therefor in the Company's consolidated balance sheet dated as of January 31, 2006 and if the Company obtains a general release in connection therewith.

(1) Indebtedness, Investments. Incur, assume, guarantee or otherwise become liable for any indebtedness for borrowed money, including, without limitation, capital leases, or make any loans or advances or make any capital contributions to, or investments in, any other Person in excess of \$250,000 individually, or \$500,000 in the aggregate, except indebtedness to trade creditors of the Company or its Subsidiaries in

the ordinary and usual course consistent with past practices. This Section 4.01(1) shall not apply to performance bonds, letters of credit and similar instruments in the conduct of its business in the ordinary course.

(m) Material Contract. Enter into, modify, amend, terminate (other than at the end of the term thereof) or waive any material rights under any Company Material Contract, other than in the ordinary and usual course of business consistent with past practice.

(n) Reorganization. Adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Company or any Subsidiary (other than the Merger).

(o) Adverse Actions. Take, or omit to take, any action which would reasonably be expected to have the legal or practical effect of materially delaying or preventing the expiration or termination of any applicable waiting period under the HSR Act prior to the Effective Time.

(p) Confidentiality Agreements. The Company will not waive any of its rights in any material respect under any Existing Confidentiality Agreement.

(q) Commitments. Enter into any contract with respect to, or otherwise agree or commit to do, directly or indirectly, any of the foregoing.

4.02 *Agreements of Parent*. Parent agrees that, from the date hereof until the Effective Time, without the prior written consent of the Company, it will not, and will cause each of its Subsidiaries not to:

(a) Adverse Actions. Take, or omit to take, any action that would reasonably be expected to result in any of the conditions to the Merger set forth in Article VII not being satisfied in a timely manner, including enter into agreements with respect to any acquisitions, mergers, consolidations or business combinations which has the effect of materially delaying or preventing the consummation of the Merger or the other Transactions or the obtaining of any regulatory or other consent or approval contemplated hereby.

(b) Commitments. Enter into any contract with respect to, or otherwise agree or commit to do, directly or indirectly, any of the foregoing.

ARTICLE V

REPRESENTATIONS AND WARRANTIES

5.01 *Disclosure Schedules*. Before entry into this Agreement, the Company delivered to Parent a schedule (the "*Disclosure Schedule*") which is divided into sections and subsections corresponding to the sections and subsections of this Agreement setting forth, among other things, items the disclosure of which is necessary or appropriate either (a) in response to an

express disclosure requirement contained in a provision hereof or (b) as an exception to one or more representations or warranties contained in Section 5.03 or to one or more of its covenants contained in Article IV; provided that the inclusion of an item in the Disclosure Schedule as an exception to a representation or warranty will not by itself be deemed an admission by the Company that such item is material or was required to be disclosed therein. Unless otherwise specifically defined therein or the context otherwise requires, capitalized terms set forth in the Disclosure Schedule shall have the meanings ascribed to such terms in this Agreement.

5.02 Standard for Breach of Representations and Warranties. For all purposes of this Agreement, no representation or warranty of the Company or Parent contained in Section 5.03 (other than the representations and warranties contained in Section 5.03(b), which shall be true in all respects except for de minimus variations, and in 5.03(g)(2)(B), which shall be true in all respects) will be deemed untrue, and no party will be deemed to have breached a representation or warranty, where such failure to be true or breach of such representation or warranty would not be material and adverse to the Company and its Subsidiaries taken as a whole, or Parent and its Subsidiaries taken as a whole, as the case may be.

5.03 *Representations and Warranties.* Except as Previously Disclosed, the Company hereby represents and warrants to Parent, and Parent hereby represents and warrants to the Company, except where specifically indicated as applicable only to the Company or Parent, as the case may be, as follows:

(a) Organization and Standing. It is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation. It is duly qualified to do business and is in good standing in all jurisdictions where its ownership or leasing of property or assets or its conduct of business requires it to be so qualified.

(b) Company Stock. In the case of the Company only:

The authorized capital stock of the Company consists of 100,000,000 shares of Company Common Stock. As of February 24, 2006, no more than 29,367,498 shares of Company Common Stock were issued and outstanding. As of February 24, 2006, no more than 2,106,550 shares of Company Common Stock were issuable upon exercise of Company Stock Options under Company Stock Plans. As of February 24, 2006, no shares of Company Common Stock were issuable upon exercise of any other Rights under Company Stock Plans. As of the date of this Agreement, (1) no shares of Company Common Stock have been issued since February 24, 2006, other than shares of Company Common Stock issued pursuant to the exercise of Company Stock Options issued and outstanding as of February 24, 2006 and (2) no Company Stock Options have been issued since February 24, 2006. The outstanding shares of Company Common Stock have been duly authorized and are validly issued and outstanding, fully paid and nonassessable, and not subject to preemptive rights (and were not issued in violation of any preemptive rights). Except as set forth above and except for shares issuable pursuant to Company Stock Plans, there are no shares of Company Stock reserved for issuance, the Company does not have any Rights outstanding with respect to Company Stock and the Company does not have any commitment to authorize, issue or sell any Company

Common Stock, preferred stock or Rights, except pursuant to this Agreement, outstanding Company Stock Options and Company Stock Plans. The Company has no commitment to redeem, repurchase or otherwise acquire any shares of Company Common Stock. The Company does not have a poison pill, shareholder rights plan or agreement or any similar plans or agreements having similar consequences that will be triggered as a result of the Transactions.

(c) Company Subsidiaries. In the case of the Company only:

(1) (A) It owns, directly or indirectly, all the outstanding equity securities of each of its Subsidiaries free and clear of any Liens, (B) no equity securities of any of its Subsidiaries are or may become required to be issued (other than to it or its wholly owned Subsidiaries) by reason of any Right or otherwise, (C) there are no contracts, commitments, understandings or arrangements by which any of such Subsidiaries is, or may be, bound to sell or otherwise transfer any equity securities of any such Subsidiaries (other than to it or its wholly-owned Subsidiaries), (D) there are no contracts, commitments, understandings or arrangements relating to its rights to vote or to dispose of such securities and (E) all the equity securities of each Subsidiary held by it or its Subsidiaries have been duly authorized and are validly issued and outstanding, fully paid and nonassessable.

(2) Each of its Subsidiaries has been duly organized and is validly existing in good standing under the laws of the jurisdiction of its organization, and is duly qualified to do business and in good standing in all jurisdictions where its ownership or leasing of property or its conduct of business requires it to be so qualified.

(d) Power. It and, in the case of the Company only, each of its Subsidiaries, has the corporate (or comparable) power and authority to carry on its business as it is now being conducted and to own all its properties and assets; and it has the corporate (or comparable) power and authority to execute, deliver and perform its obligations under this Agreement and to consummate the Transactions.

(e) Authority. It has duly authorized, executed and delivered this Agreement. Subject only in the case of the Company to receipt of the affirmative vote of the holders of at least two-thirds of the outstanding shares of Company Common Stock to approve the Plan of Merger (the "*Company Shareholder Approval*"), this Agreement and the Transactions have been authorized by all necessary respective corporate action. In the case of the Company only, at a meeting duly called and held, the Company Board has (1) determined that this Agreement and the transactions contemplated hereby, including the Merger (collectively, the "*Transactions*"), are fair to, and in the best interests of, the holders of Shares, (2) approved, adopted and declared advisable this Agreement and the Transactions (such approval and adoption having been made in accordance with the TBCA, including, without limitation, Part 13 thereof), and (3) recommended that the Merger and this Agreement be approved by the shareholders of the Company (the "*Company Board Recommendation*"); provided, however, that any withdrawal,

modification or qualification of such recommendation in accordance with Section 6.06 shall not be deemed a breach of this Section 5.03(e). This Agreement is its valid and legally binding obligation, enforceable in accordance with its terms (except as enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and similar laws of general applicability relating to or affecting creditors' rights or by general equity principles).

(f) Regulatory Approvals; No Defaults.

(1) No consents or approvals of, or filings or registrations with, any Governmental Authority or with any third party are required to be made or obtained by it or, in the case of the Company only, any of its Subsidiaries in connection with the execution, delivery or performance by it of this Agreement or to consummate the Merger, except for (A) filings of applications and notices with, receipt of approvals or non-objections from, and expiration of related waiting periods, required by U.S. federal or foreign Governmental Authorities, (B) filing of notices, and expiration of the related waiting period, under the HSR Act, (C) filings as may be required by the Securities Act or the Exchange Act including, in the case of the Company only, filing of the Proxy Statement with the SEC, (D) receipt of the Company Shareholder Approval, and (E) the filing of the Articles of Merger and the issuance of the Certificate of Merger.

(2) Subject to receipt of the consents and approvals referred to in the preceding paragraph, and the expiration of related waiting periods, and required filings with U.S. federal or foreign Governmental Authorities, the execution, delivery and performance of this Agreement and the consummation of the Transactions do not and will not (A) constitute a breach or violation of, or a default under, or give rise, in the case of the Company only, to any Lien or any acceleration of remedies, penalty, increase in material benefit payable or right of termination under, any law, rule or regulation or any judgment, decree, order, governmental permit or license, or agreement, indenture or instrument of it or, in the case of the Company only, of any of its Subsidiaries or to which it or, in the case of the Company only, any of its Subsidiaries or properties is subject or bound, (B) constitute a breach or violation of, or a default under, rule, regulation, judgment, decree, order, governmental permit or license, agreement, indenture or instrument of permit or license, agreement, indenture or approval under any such law, rule, regulation, judgment, decree, order, governmental permit or license, agreement, indenture or instrument or license, agreement, indenture or instrument.

(g) Company Financial Reports and Regulatory Filings; Material Adverse Effect. In the case of the Company only:

(1) Since January 31, 2001, the Company has filed with the SEC all material forms, statements, reports and documents required to be filed by it under the Exchange Act and the Securities Act. Its Annual Reports on Form 10-K for the fiscal years ended January 31, 2003, 2004, and 2005 and all other reports, registration statements, definitive proxy statements or information statements filed by it or any of its Subsidiaries subsequent to January 31, 2002 under the

Securities Act, or under Section 13(a), 13(c), 14 or 15(d) of the Exchange Act, in the form filed (collectively, its "*Regulatory Filings*") with the SEC as of the date filed, (A) complied in all material respects as to form with the applicable requirements under the Securities Act or the Exchange Act, as the case may be, and (B) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading; and each of the consolidated statements of financial position contained in or incorporated by reference into any such Regulatory Filing (including the related notes and schedules) fairly presented in all material respects the financial position of it and its Subsidiaries on a consolidated basis as of the date of such statement in accordance with GAAP consistently applied during the periods involved, except in each case as may be noted therein, and, in the case of quarterly reports filed on Form 10-Q, subject to normal year-end audit adjustments as permitted by SEC Form 10-Q promulgated under the Exchange Act in the case of unaudited statements.

(2) Since January 31, 2005, except for liabilities (A) permitted or contemplated by this Agreement, (B) incurred in the ordinary course of business, or (C) that would not reasonably be expected to be material and adverse to the Company and its Subsidiaries taken as a whole, the Company has not incurred any liabilities of a nature required to be set forth or reflected in a balance sheet prepared in accordance with GAAP.

(3) Since January 31, 2005, (A) it and its Subsidiaries have conducted their respective businesses in the ordinary and usual course consistent with past practice (excluding the incurrence of expenses related to this Agreement and the Transactions) and (B) no event has occurred or circumstance has arisen that has had or is reasonably likely to have a Material Adverse Effect with respect to it.

(h) Company Litigation. In the case of the Company only, as of the date hereof, there is no suit, action, investigation or proceeding pending or, to its Knowledge, threatened against or affecting it or any of its Subsidiaries, nor is there any judgment, decree, injunction, rule or order of any Governmental Authority or arbitration outstanding against it or any of its Subsidiaries.

(i) Company Regulatory Matters. In the case of the Company only, neither it nor any of its Subsidiaries is subject to or has been advised by any Governmental Authority that it is likely to become subject to, any written order, decree or similar agreement from or with any U.S. federal Governmental Authority charged with the regulation of, or enforcement of matters concerning, persons engaged in Defense Businesses.

(j) Company Compliance with Laws. In the case of the Company only, it and each of its Subsidiaries:



(1) conducts its business in compliance with all applicable federal, state, local and foreign statutes, laws, regulations, ordinances, rules, judgments, orders or decrees applicable thereto or to the employees conducting such businesses;

(2) has all permits, licenses, authorizations, orders and approvals of, and has made all filings, applications and registrations with, all Governmental Authorities that are required in order to permit them to own or lease their properties and to conduct their businesses as presently conducted; all such permits, licenses, certificates of authority, orders and approvals are in full force and effect and, to its Knowledge, no suspension or cancellation of any of them is threatened; and

(3) has not received, since January 31, 2005, written notification from any Governmental Authority (A) asserting that it or any of its Subsidiaries is not in compliance with any of the statutes, regulations or ordinances which such Governmental Authority enforces or (B) threatening to revoke any license, franchise, permit or governmental authorization.

(k) Company Material Contracts; Defaults. In the case of the Company only:

(1) Neither it nor any of its Subsidiaries is a party to, bound by or subject to any agreement, contract, arrangement, commitment or understanding (whether written or oral) (A) that is a "material contract" within the meaning of Item 601(b)(10) of the SEC's Regulation S-K or (B) that restricts in any material respect the conduct of business by it or any of its Subsidiaries or its or their ability to compete in any line of business (each a "*Material Contract*").

(2) Neither it nor any of its Subsidiaries is in default under any Material Contract, and to its Knowledge, there has not occurred any event that, with the lapse of time or the giving of notice or both, would constitute such a default.

(l) Company Taxes. In the case of the Company only,

(i) (1) all Tax Returns that are required to be filed (taking into account any extensions of time within which to file) by or with respect to it and its Subsidiaries have been duly and timely filed, (2) all Taxes shown to be due on the Tax Returns referred to in clause (1) have been paid in full, (3) all Taxes that it or any of its Subsidiaries is obligated to withhold from amounts owing to any employee, creditor or third party have been withheld and paid over to the proper Governmental Authority, to the extent due and payable, and (4) no extensions or waivers of statutes of limitation have been given by or requested with respect to any of its U.S. federal income taxes or those of its Subsidiaries. The Company has either paid or made provision in accordance with GAAP, in the consolidated financial statements included in the Regulatory Filings filed before the

date hereof, for all Taxes of the Company and its Subsidiaries that accrued on or before the end of the most recent period covered by its Regulatory Filings filed before the date hereof. No Liens for Taxes exist with respect to any of its assets or properties or those of its Subsidiaries.

(ii) The Company's reserve for Taxes contained in its consolidated financial statements included in the Regulatory Filings filed before the date hereof is adequate, as determined under GAAP, to cover any unpaid Taxes of the Company and its Subsidiaries.

(iii) Neither the Company nor any of its Subsidiaries is a party to or bound by any Tax allocation or sharing agreement which could require a payment by the Company or a Subsidiary to a party other than the Company or a Subsidiary. Neither the Company nor any of its Subsidiaries has any liability for the Taxes of any Person (other than Company or any of its Subsidiaries) under Reg. §1.1502-6 (or any similar provision of state, local, or foreign law).

(iv) No written claim has been made by a taxing authority in a jurisdiction where the Company or its Subsidiaries does not file Tax Returns that the Company or its Subsidiaries may be subject to taxation by that jurisdiction.

(v) Neither the Company nor any of its Subsidiaries is a party to any agreement, contract or arrangement or plan that could result, separately or in the aggregate, as a result of the transactions contemplated by this Agreement, in the payment of any "excess parachute payment" within the meaning of Code Section 280G (or any corresponding provision of state, local or foreign Tax law).

(vi) Within the past two years (or any other Tax year for which the applicable statute of limitations is open as of the date of this Agreement), neither the Company nor any of its Subsidiaries has distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Code Section 355 or Code Section 361.

(vii) Neither the Company nor any of its Subsidiaries will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any "closing agreement" as described in Code Section 7121 (or any corresponding provision).

(m) Company Employee Benefit Plans. In the case of the Company only:

(1) All of its Benefit Arrangements are Previously Disclosed. True and complete copies of all Benefit Arrangements, including, but not limited to,

any trust instruments and insurance contracts forming a part of any Benefit Arrangements, and all amendments thereto, have been made available to Parent.

(2) All of its Benefit Arrangements, other than "multiemployer plans" within the meaning of Section 3(37) of ERISA, are in compliance with ERISA, the U.S. Internal Revenue Code of 1986 (the "*Code*") and other applicable laws. Each of its Benefit Arrangements which is an "employee pension benefit plan" within the meaning of Section 3(2) of ERISA ("*Pension Plan*"), and which is intended to be qualified under Section 401(a) of the Code, has received a favorable determination letter from the U.S. Internal Revenue Service, and to its Knowledge there are no circumstances reasonably likely to result in revocation of any such favorable determination letter or which would result in material costs under the U.S.-Internal Revenue Service's Employee Plans Compliance Resolution System.

(3) None of it, any of its Subsidiaries or any entity which is considered one employer with it under Section 4001 of ERISA or Section 414 of the Code (an "*ERISA Affiliate*") has contributed to a "multiemployer plan", within the meaning of Section 3(37) of ERISA, at any time within the last six years.

(4) All contributions required to be made under the terms of any of its Benefit Arrangements have been timely made or have been reflected on its consolidated financial statements included in its Regulatory Filings. None of its Pension Plans or any single-employer plan of any of its ERISA Affiliates has an "accumulated funding deficiency" (whether or not waived) within the meaning of Section 412 of the Code or Section 302 of ERISA and none of its ERISA Affiliates has an outstanding funding waiver. Neither it nor any of its Subsidiaries has provided, or is required to provide, security to any of its Pension Plans or to any single-employer plan of any of its ERISA Affiliates pursuant to Section 401(a)(29) of the Code.

(5) Neither it nor any of its Subsidiaries has any obligations for retiree health and life benefits under any Benefit Arrangement or collective bargaining agreement, other than for benefits required by applicable law or the cost of which is borne by the participant.

(6) Neither its execution of this Agreement, the consummation of the Transactions nor the Company Shareholder Approval will, either alone or in conjunction with another event (such as termination of employment), (A) entitle any of its employees or any employees of its Subsidiaries to severance pay or any increase in severance pay or (B) accelerate the time of payment or vesting or trigger any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or trigger any other material obligation pursuant to, any of its Benefit Arrangements.

(n) Takeover Laws and Provisions Applicable to the Company. In the case of the Company only, it has taken all action required to be taken by it in order to exempt this

Agreement and the Transactions from, and this Agreement and the Transactions are exempt from, the provisions of Article 13.03 of the TBCA, and the requirements of any "moratorium," "control share", "fair price", "affiliate transaction", "business combination" or other anti-takeover laws and regulations of any state (collectively, "*Takeover Laws*"). In the case of the Company only, it has taken all action required to be taken by it in order to make this Agreement and the Transactions comply with, and this Agreement and the Transactions do comply with, the requirements of any Articles, Sections or provisions of its Constituent Documents concerning "business combination", "fair price", "voting requirement", "constituency requirement" or other related provisions (collectively, "*Takeover Provisions*").

(o) Financial Advisors and Consultants. None of it, its Subsidiaries or any of their officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finder's fees in connection with the Transactions, except that, in connection with this Agreement, Parent has retained Wachovia Securities and Pruzan & Co. LLC as its financial advisors, and the Company has retained Merrill Lynch & Co. as its financial advisor and Robinson Partners as an outside consultant, the arrangements with which have been disclosed to Parent prior to the date hereof.

(p) Sarbanes-Oxley Act. In the case of the Company only, it is in compliance with the provisions of the Sarbanes-Oxley Act, including Section 404 thereof, and to its Knowledge, the certifications provided pursuant to Sections 302 and 906 thereof with each Regulatory Filing of the Company, at the time of filing or submission of each such certification, were accurate.

(q) Company Labor Matters. In the case of the Company only, neither it nor any of its Subsidiaries is a party to, or is bound by, any collective bargaining agreement, contract or other agreement or understanding with a labor union or labor organization, nor is it or any of its Subsidiaries the subject of a proceeding asserting that it or any such Subsidiary has committed an unfair labor practice (within the meaning of the National Labor Relations Act) or seeking to compel it or such Subsidiary to bargain with any labor organization as to wages and conditions of employment, nor is there any strike or other labor dispute involving it or any of its Subsidiaries pending or, to its Knowledge, threatened, nor is it aware, as of the date of this Agreement, of any activity involving it or any of its Subsidiaries' employees seeking to certify a collective bargaining unit or engaging in any other organization activity.

(r) Company Environmental Matters. In the case of the Company only, there are no proceedings, claims, actions or investigations of any kind pending or, to the Knowledge of the Company, threatened, against the Company or any Subsidiary in any court, agency or other Governmental Authority or in any arbitral body, arising under any Environmental Law; the Company and its Subsidiaries have conducted their respective operations in compliance with their respective environmental permits and the limitations, restrictions, conditions, standards, prohibitions, requirements and obligations of applicable Environmental Laws; all permits required under Environmental Laws have been obtained and are in full force and effect; there have been no releases of Materials of

Environmental Concern or other conditions at any property owned, or operated by the Company or its Subsidiaries that would require remediation under Environmental Laws; and, no facts or circumstances exist that would reasonably be expected to result in any liabilities under Environmental Law or common law relating to the operations of the Company or its Subsidiaries as a result of (1) any violation by the Company or its Subsidiaries of any Environmental Law, or (2) the release, storage, handling, treatment, transportation or disposal of any Materials of Environmental Concern generated by the Company's or its Subsidiaries' operations.

(s) Knowledge Regarding Representations; Satisfaction of Conditions; No Additional Representations. In the case of Parent only:

(1) As of the date hereof, neither Parent nor Merger Sub is aware of any inaccuracy or misstatement in, or breach of, any representation or warranty of the Company contained herein.

(2) As of the date hereof, neither Parent nor Merger Sub is aware of any reason why the conditions set forth in Article VII hereof would not be timely satisfied.

(3) Parent acknowledges that it and its representatives have received access to such books and records, facilities, equipment, contracts and other assets of the Company and its Subsidiaries which it and its representatives have desired or requested to review, and that it and its representatives have had full opportunity to meet with the management of the Company to discuss the businesses and assets of the Company and its Subsidiaries. Parent acknowledges that neither the Company nor any other person has made any representation or warranty, expressed or implied, as to the accuracy or completeness of any information regarding the Company furnished or made available to Parent and its representatives except as expressly set forth in Section 5.03 hereof, and neither the Company nor any other person shall have or be subject to any liability to Parent, Merger Sub or any other person resulting from the Company's information memorandum dated December 2005, as subsequently updated, supplemented or amended (the "Information Memorandum") or any information, documents or material made available to Parent in the due diligence materials provided to Parent, including in the "data room," other management presentations (formal or informal including the management presentation dated January 2006 and entitled "Evaluation Materials" (the "Management Presentation")) or in any other form in connection with the Transactions. Without limiting the foregoing, the Company makes no representation or warranty to Parent or Merger Sub with respect to (a) the information set forth in the Information Memorandum or the Management Presentation or (b) any financial projection or forecast relating to the Company or its Subsidiaries, whether or not included in the Information Memorandum or the Management Presentation. Except as otherwise expressly provided herein, no information or knowledge obtained by any party as described in this Section 5.03(s)(3) or otherwise shall affect or be deemed to modify any representation or

warranty made by any party hereunder, unless and to the extent a party has actual knowledge to the contrary.

(t) Proxy Statement. In the case of the Company only:

(1) The proxy statement to be sent to the shareholders of the Company in connection with such Shareholders' Meeting (such proxy statement, as amended or supplemented, being referred to herein as the "*Proxy Statement*"), shall not, at the date the Proxy Statement (or any amendment or supplement thereto) is first mailed to shareholders of the Company or at the time of the Shareholders' Meeting, contain any statement which, at the time and in light of the circumstances under which it was made, be false or misleading with respect to any material fact, or omit to state any material fact necessary in order to make the statements therein not false or misleading or necessary to correct any statement in any earlier communication with respect to the solicitation of proxies for the Shareholders' Meeting which shall have become false or misleading.

(2) Notwithstanding the foregoing, the Company makes no representation or warranty with respect to any information supplied by Parent, Merger Sub or any of Parent's or Merger Sub's representatives for inclusion in the Proxy Statement.

(3) The Proxy Statement shall comply in all material respects as to form with the requirements of the Exchange Act and the rules and regulations thereunder.

(4) Merrill Lynch & Co. has consented to the inclusion of the full text of its opinion described in Section 5.03(o) in the Proxy Statement and the Company has provided a copy of such opinion to the Parent and Merger Sub.

(u) Proxy Statement. In the case of Parent and Merger Sub only:

(1) The information supplied by Parent for inclusion in the Proxy Statement shall not, at the date the Proxy Statement (or any amendment or supplement thereto) is first mailed to shareholders of the Company or at the time of the Shareholders' Meeting, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading, or necessary to correct any statement in any earlier communication with respect to the solicitation of proxies for the Shareholders' Meeting which shall have become false or misleading.

(2) Notwithstanding the foregoing, Parent and Merger Sub make no representation or warranty with respect to any information supplied by the Company or any of its representatives for inclusion in the Proxy Statement.

(v) Required Vote of Company Shareholders. In the case of the Company only, the affirmative vote of the holders of at least two-thirds of the outstanding Shares,

voting together as a single class, is the vote required to approve the Merger and this Agreement, and no other vote of the security holders of the Company is required by law, the Company's Constituent Documents or otherwise in order for the Company to consummate the Merger and the other Transactions contemplated hereby.

(w) Confidentiality Agreements. In the case of the Company only, each confidentiality agreement executed by any person participating in the auction process conducted by the Company in connection with its solicitation of acquisition offers (the "*Existing Confidentiality Agreements*") is substantially similar to the Confidentiality Agreement executed by Parent, including, but not limited to, with respect to terms concerning confidentiality, "standstill" obligations relating to the purchase or offer to purchase any equity securities of the Company and non-solicitation of the Company's employees. In addition, the Company has not waived any of its rights under any of such other confidentiality agreements.

5.04 *Representations and Warranties about Merger Sub.* Parent and Merger Sub hereby jointly and severally represent and warrant to the Company as follows:

(a) Organization and Standing. Merger Sub is a corporation duly organized, validly existing and in good standing under the laws of the State of Texas. It is duly qualified to do business and is in good standing in all jurisdictions where its ownership or leasing of property or assets or its conduct of business requires it to be so qualified.

(b) Merger Sub Stock. The authorized capital stock of Merger Sub consists of 1,000 shares of Merger Sub Common Stock. All of the issued and outstanding capital stock of Merger Sub is owned by Parent as its sole shareholder. The outstanding shares of Merger Sub Common Stock have been duly authorized and are validly issued and outstanding, fully paid and nonassessable, and not subject to preemptive rights (and were not issued in violation of any preemptive rights).

(c) Power. Merger Sub has the corporate power and authority to carry on its business as it is now being conducted and to own all its properties and assets, and it has the corporate power and authority to execute, deliver and perform its obligations under this Agreement and to consummate the Transactions.

(d) Authority. Merger Sub has duly authorized, executed and delivered this Agreement, and this Agreement and the Transactions have been authorized by all other necessary corporate actions, including the necessary approval of Parent as the sole shareholder of Merger Sub. This Agreement is its valid and legally binding obligation, enforceable in accordance with its terms (except as enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and similar laws of general applicability relating to or affecting creditors' rights or by general equity principles).

(e) Funds. Merger Sub has, or will have at the Effective Time, sufficient funds available to satisfy the obligation to pay the Per Share Consideration in the Merger.

ARTICLE VI COVENANTS

6.01 *Reasonable Best Efforts.* Subject to the terms and conditions of this Agreement, Parent, Merger Sub and the Company will each use its reasonable best efforts to take, or cause to be taken, in good faith, all actions, and to do, or cause to be done, all things necessary, proper or desirable, or advisable under applicable laws, so as to permit consummation of the Merger as promptly as practicable and otherwise to enable consummation of the Transactions, including, in the case of the Company, using its reasonable best efforts to obtain the consents of the Governmental Authorities that are parties to the contracts set forth on Schedule 6.01 under those contracts, including the giving of any notices thereunder, and each will cooperate fully with, and furnish information to, the other party to that end. In furtherance and not in limitation of the covenants of the parties contained in this Section 6.01, if any objections are asserted with respect to the Transactions under any antitrust law, or if any suit is threatened to be instituted, by any Governmental Authority or any private party challenging any of the Transactions or otherwise brought under any antitrust law or other applicable law that would otherwise prohibit or materially impair or materially delay the consummation of the Transactions, each of Parent, Merger Sub and the Company shall use its reasonable best efforts to resolve any such objections or suits so as to permit consummation of the Transactions as soon as practicable, and in any event before the Termination Date. Notwithstanding anything in this Section 6.01 to the contrary, Parent shall not be required to sell or dispose of any businesses or assets, to the extent any such sale or disposition would have a material adverse effect on Parent and the Company, taken as a whole.

6.02 *Shareholder Approvals.* The Company, acting through the Company Board, shall, in accordance with applicable law, including the TBCA, and the Company's Articles of Incorporation and By-laws, duly call, give notice of, convene and hold an annual or special meeting of its shareholders as promptly as practicable following the date hereof for the purpose of considering and taking action on this Agreement and the Merger (the "*Shareholders' Meeting*"). At the Shareholders' Meeting, Parent and Merger Sub shall cause all Shares then owned by them and their subsidiaries to be voted in favor of the adoption of this Agreement.

6.03 *Proxy Statement*. As promptly as practicable after the date hereof, the Company shall file the Proxy Statement with the SEC under the Exchange Act, and shall use its reasonable best efforts to have the Proxy Statement cleared by the SEC promptly. The Company shall provide Parent and its counsel a reasonable opportunity to review and comment upon the Proxy Statement prior to the filing with the SEC. In addition, the Company shall provide Parent and its counsel a written copy of any comments the Company or its counsel may receive from the SEC or its staff with respect to the Proxy Statement promptly after receipt of such comments and with copies of any written responses to such comments and telephonic notification of any verbal responses to such comments by the Company or its counsel. No filing of, or amendment to, or written correspondence to the SEC or its staff with respect to the Proxy Statement thereon. Each of the Company, Parent and Merger Sub agrees to use its reasonable best efforts, after consultation with the other parties hereto, to respond promptly to all such comments of and requests by the SEC and to cause the Proxy Statement and all required

amendments and supplements thereto to be mailed to the holders of Shares entitled to vote at the Shareholders' Meeting at the earliest practicable time. Subject in each case to Section 6.06, the Company agrees to use its reasonable best efforts to solicit from the holders of Shares proxies in favor of the adoption of this Agreement and the Merger and take all other action necessary or advisable to secure, at the Shareholders' Meeting the Company Shareholder Approval, by the vote described in Section 5.03(v) of this Agreement.

6.04 *Press Releases.* Parent and the Company will consult with each other before issuing any press release with respect to the Merger or this Agreement and will not issue any such press release without the prior written consent of the other party, which will not be unreasonably withheld or delayed; provided that a party may, without the prior consent of the other party (but after prior consultation, to the extent practicable in the circumstances), issue any such press release or a public statement or file any required documents with the SEC as may be required by applicable law or securities exchange rules. Parent and the Company will cooperate to develop all public communications and make appropriate members of management available at presentations related to the Transactions as reasonably requested by the other party.

6.05 Access; Information.

(a) Company agrees that upon reasonable notice and subject to applicable laws relating to the exchange of information, it will (and will cause its Subsidiaries to) afford Parent, and Parent's Representatives and accountants, such access during normal business hours throughout the period before the Effective Time to the books, records (including, without limitation, Tax Returns and work papers of independent auditors) and properties and to such other information as Parent may reasonably request; provided, however, that such access shall not unreasonably disrupt the operations of the Company. All requests for such access shall be made to representatives of the Company as the Company shall designate, who shall be solely responsible for coordinating all such requests and all access permitted hereunder. Neither Parent nor any of Parent's Representatives shall contact any of the employees, customers or suppliers of the Company in connection with the Transactions, whether in person or by telephone, mail or other means of communication, without the specific prior written authorization of such representatives of the Company may designate. The Company will not be required to afford access or disclose information that would jeopardize attorney-client privilege, contravene any binding agreement with any third party or violate any law or regulation. The parties will make reasonable appropriate substitute arrangements in circumstances where the previous sentence applies.

(b) Each party will hold any information provided in connection with this Agreement or the Transactions confidential in accordance with the Confidentiality Agreement.

6.06 No Solicitation.

(a) From the date hereof until the Effective Time or the earlier termination of this Agreement, the Company shall not, directly or indirectly, through any officer, director, employee, representative or agent of the Company or any of its Subsidiaries

(including any investment banker, attorney or accountant retained by it or any of its Subsidiaries) (collectively, the "Company Representatives"), and shall use commercially reasonable efforts to cause the Company Representatives, not to, directly or indirectly, (i) initiate, solicit, or knowingly encourage or facilitate inquiries or proposals with respect to, or participate in negotiations concerning, or provide any confidential information or data to, or have any substantive discussions with, any third party relating to or that would reasonably be expected to lead to, any Acquisition Proposal (except to notify such person as to the existence of the provisions of this Section 6.06), or (ii) enter into any merger agreement, acquisition agreement, letter of intent or similar document providing for or otherwise relating to any Acquisition Proposal; provided, that notwithstanding anything in this Agreement to the contrary, the Company may, directly or indirectly through any of the Company Representatives, furnish information and data to, and participate in discussions or negotiations with, any person that has made a bona fide written Acquisition Proposal if, and only to the extent that, (I) (A) such Acquisition Proposal is at a price higher than the Per Share Consideration and the Company's Board believes in good faith that there is a substantial likelihood that the consummation of such Acquisition Proposal is probable and (B) after the receipt of advice from outside legal counsel, the Company Board determines in good faith that the failure to take such action would be inconsistent with its fiduciary obligations to the Company's shareholders under applicable law, (II) the Company shall not have violated any of the provisions of this Section 6.06(a), and (III) prior to taking such action, the Company receives from such person an executed confidentiality agreement on terms with respect to confidentiality, standstill obligations relating to the purchase or offer to purchase any equity securities of the Company and non-solicitation of the Company's employees on substantially similar terms to the Confidentiality Agreement; provided, that, notwithstanding anything herein to the contrary, the Company and its Representatives may take, and shall not be prohibited or otherwise limited by the terms of such confidentiality agreement in taking, any action that is otherwise contemplated by this Section 6.06, including the waiver by the Company of the standstill provision to permit discussions and negotiations, the entering into of a definitive agreement and any other actions in the furtherance thereof with respect to such Acquisition Proposal.

(b) Notwithstanding anything herein to the contrary, in the event of any bona fide written Acquisition Proposal that has not been withdrawn, (i) the Company Board may approve or recommend, and the Company may enter into a definitive agreement providing for, an Acquisition Proposal that is submitted to the Company prior to the Shareholders' Meeting, and (ii) the Company Board may withdraw or modify in any manner adverse to Parent the Company Board Recommendation (a "*Company Board Change of Recommendation*"), if (I) the Company shall not have violated any of the provisions of Section 6.06(a), (II) the Company Board will take any such action, during which three-business Days prior to any meeting of the Company Board at which the Company Board will take any such action, during which three-business day period Parent may propose revisions to the terms of this Agreement, (III) notwithstanding such revisions proposed by Parent pursuant to clause (II) above, the Company Board makes the determination that such Acquisition Proposal constitutes a Superior Proposal, (IV) after the receipt of advice from outside legal counsel, the Company Board determines in

good faith that the failure to take such action would be inconsistent with its fiduciary obligations to the Company's shareholders under applicable law.

(c) The Company shall promptly (and in any event within 24 hours of receipt thereof) notify Parent after receipt of any Acquisition Proposal (which notice shall identify the person making such Acquisition Proposal and set forth the material terms thereof), any material modification of or material amendment to any Acquisition Proposal (including the material terms thereof), or any request for non-public information relating to the Company by any person that has notified the Company of its intention to make any Acquisition Proposal and the terms and conditions of such request.

(d) The Company shall immediately cease and cause to be terminated any existing discussions or negotiations with any persons (other than Parent) conducted heretofore with respect to any of the foregoing. The Company shall immediately request each person who has heretofore executed a confidentiality agreement in connection with its consideration of acquiring Company or any portion thereof to return or destroy in accordance with the terms of such confidentiality agreement all confidential information heretofore furnished to such person by or on behalf of Company.

(e) Nothing contained in this Agreement shall prohibit the Company or the Company Board from taking and disclosing to its shareholders a position contemplated by Rule 14e-2(a) promulgated under the Exchange Act or from making any disclosure to the Company's shareholders if, in the good faith judgment of the Company Board, after consultation with outside counsel, failure to take such action would be inconsistent with applicable law.

(f) Nothing in this Agreement shall prohibit or restrict the Company Board from making a Company Board Change of Recommendation to the extent that the Company Board determines in good faith, following consultation with outside legal counsel, that failure to take such action would constitute a breach of the Company Board's fiduciary obligations under applicable law.

6.07 *Takeover Laws and Provisions.* No party will take any action that would cause the Transactions to be subject to requirements imposed by any Takeover Law and each of them will take all necessary steps within its control to exempt (or ensure the continued exemption of) such Transactions from, or if necessary challenge the validity or applicability of, any applicable Takeover Law, as now or hereafter in effect. No party will take any action that would cause the Transactions not to comply with any Takeover Provisions and each of them will take all necessary steps within its control to make such Transactions comply with (or continue to comply with) the Takeover Provisions. Prior to the Effective Time, the Company shall not elect to adopt, or otherwise take action to become subject to, the TBOC.

6.08 Regulatory Applications.

(a) Parent and the Company and their respective Subsidiaries will cooperate and use all reasonable best efforts to prepare as promptly as possible all documentation, to effect all filings and to obtain all material permits, consents, approvals and

authorizations of all third parties and Governmental Authorities necessary to consummate the Transactions (the "*Requisite Approvals*") as promptly as practicable, and will make all necessary filings in respect of those Requisite Approvals as soon as practicable. Each of Parent and the Company will have the right to review in advance, and to the extent practicable each will consult with the other, in each case subject to applicable laws relating to the exchange of information, with respect to all material written information submitted to any third party or any Governmental Authority in connection with the Requisite Approvals. In exercising the foregoing right, each of the parties will act reasonably and as promptly as practicable. Each party agrees that it will consult with the other party with respect to obtaining all Requisite Approvals and each party will keep the other party apprised of the status of material matters relating to completion of the Transactions.

(b) Parent and the Company will, upon request, furnish the other party with all information concerning itself, its Subsidiaries, directors, officers and shareholders and such other matters as may be reasonably necessary or advisable in connection with any filing, notice or application made by or on behalf of such other party or any of its Subsidiaries with or to any third party or Governmental Authority in connection with the Transactions and to the extent permitted by the applicable Governmental Authority or any person objecting to the Transactions, give the other party the opportunity to attend and participate in any meetings and conferences with such Governmental Authority or person objecting to the Transactions.

6.09 Indemnification.

(a) The indemnification provisions of the Constituent Documents of the Surviving Corporation as in effect at the Effective Time shall not be amended, repealed or otherwise modified for a period of six years from the Effective Time in any manner that would adversely affect the rights thereunder of individuals who at the Effective Time were directors, officers or employees of the Company unless such modification is, in the opinion of outside counsel, required by applicable law. Parent shall assume, be jointly and severally liable for, and honor, guarantee and stand surety for, and shall cause the Surviving Corporation to honor, in accordance with their respective terms, each of the covenants contained in this Section 6.09.

(b) Without limiting Section 6.09(a), following the Effective Time, Parent and the Surviving Corporation will indemnify, defend and hold harmless the present and former directors, officers and employees of the Company and its Subsidiaries (each, an "*Indemnified Party*") against all costs or expenses (including reasonable attorneys' fees and costs of investigation), judgments, fines, losses, claims, damages or liabilities as incurred, in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of actions or omissions before, or at the Effective Time (including as to, or arising out of or pertaining to, the Transactions), to the fullest extent permitted by applicable law and the Constituent Documents. Parent shall pay all costs and expenses in connection with such indemnification promptly as statements therefor are received, provided, however, the Parent or the Surviving Corporation may condition the payment of any costs and

expenses upon the execution and delivery to the Parent or the Surviving Corporation of an undertaking in customary form providing solely that the Indemnified Party undertakes to repay any payments advanced to the extent that it is ultimately and finally judicially determined that the Indemnified Party is not entitled to be indemnified by the Parent or the Surviving Corporation. At and as of the Effective Time Parent shall cause the By-laws of the Surviving Corporation to conform with the Constituent Documents of the Company and with its obligations under this Section 6.09.

(c) For a period of six years following the Effective Time, Parent will cause to be maintained in effect director's and officer's liability insurance that serves to reimburse the present and former officers and directors (determined as of the Effective Time) of the Company or any of their respective Subsidiaries (as opposed to reimbursing the Company or such Subsidiary) with respect to claims against such directors and officers arising from facts or events occurring before, or at the Effective Time (including as to, arising out of or pertaining to, the Transactions), which insurance will contain at least the same coverage and amounts of coverage and will contain terms and conditions no less advantageous to the Indemnified Party than those provided in the director's and officer's liability insurance currently provided by the Company; provided, that officers and directors of the Company or any Subsidiary may be required to make application and provide customary representations and warranties to Parent's insurance carrier for the purpose of obtaining insurance required to be provided by Parent pursuant to this Section 6.09(c); provided, however, that the premium for such coverage does not exceed 200% of current premium, in which event the obligation shall be only to provide the maximum coverage available for a premium not in excess of such amount; and provided, further, the obligation of Parent, Merger Sub and the Surviving Corporation in this Section 6.09(c) shall be satisfied by the purchase by the Company on or before the Effective Time of a "tail-coverage" or "run-off" insurance policy providing the coverage required under this Section 6.09(c) for the required term hereof that is substantially in accord with the requirements of this Section 6.09(c), the premium for which does not exceed 200% of current premium.

(d) Any Indemnified Party wishing to claim indemnification under Section 6.09(a), upon learning of any claim, action, suit, proceeding or investigation described above, will promptly notify Parent; provided that failure so to notify will not affect the obligations of Parent under Section 6.09(a) unless and to the extent that Parent is actually prejudiced as a consequence.

(e) If Parent or any of its successors or assigns (1) consolidates with or merges into any other entity and is not the continuing or surviving entity of such consolidation or merger or (2) transfers all or substantially all of its assets to any other entity, then and in each case, Parent will, as a condition precedent to the consummation of any such transaction, cause proper provision to be made so that the successors and assigns of Parent will, by specificity, assume the obligations set forth in this Section 6.09 except to the extent that with respect to clause (1) such obligations shall have been assumed by operation of law.

(f) The provisions of this Section 6.09 shall survive the Effective Time and are intended to be for the benefit of, and will be enforceable by, each Indemnified Party and his or her heirs and legal representatives.

(g) Parent shall pay all reasonable expenses, including reasonable attorneys' fees and costs of investigation, that may be incurred by any Indemnified Party in enforcing all obligations provided in this Section 6.09, provided, however, Parent or the Surviving Corporation may condition the payment of any costs and expenses upon the execution and delivery to Parent or the Surviving Corporation of an undertaking in customary form providing solely that the Indemnified Party undertakes to repay any payments advanced (including legal fees and expenses) to the extent that it is ultimately and finally judicially determined that the Indemnified Party is not entitled to be indemnified by Parent or the Surviving Corporation.

6.10 Employee Matters.

(a) From the Effective Time until the date which is six months after the Closing Date (such date being referred to herein as the "*Benefits Transition Date*"), Parent shall provide the employees and former employees of the Company and its Subsidiaries, as of the Effective Time (the "*Covered Employees*") taken as a whole with employee benefits and compensation plans, programs and arrangements (including base salary and annual bonus opportunities, but specifically excluding equity grants and any Supplemental Executive Retirement Plan benefits) no less favorable, in the aggregate, than those provided by the Company or its Subsidiaries, as the case may be, to such Covered Employees immediately prior to the Effective Time. From and after the Benefits Transition Date, Parent shall provide the Covered Employees taken as a whole with employee benefits and compensation plans, programs and arrangements and severance benefits that are no less favorable, in the aggregate, to those provided to similarly situated employees of Parent and its Subsidiaries. In addition, during the 24 months after the Effective Time, a Covered Employee whose employment is terminated "without cause" (as defined in such policy) shall be entitled to receive severance payments and benefits under the Company's severance policy described in the Disclosure Schedule.

(b) From and after the Effective Time, Parent shall, (1) provide all Covered Employees with service credit (for purposes of eligibility (including eligibility for retirement), participation, vesting, levels of benefits and benefit accruals under any employee benefit or compensation plan, program or arrangement adopted, maintained or contributed to by Parent or any of its Subsidiaries in which Covered Employees are eligible to participate, but specifically excluding any Supplemental Executive Retirement Plan benefits), for all periods of employment with the Company or any of its Subsidiaries (or their predecessor entities) prior to the Effective Time to the extent credited by the Company for purposes of a comparable plan (provided that there will be no duplication of benefits) and (2) cause any pre-existing conditions, limitations, eligibility waiting periods or required physical examinations under any group health plans of Parent or any of its Subsidiaries to be waived with respect to the Covered Employees and their eligible dependents to the extent waived under the corresponding plan (for a comparable level of

coverage) in which the applicable Covered Employee participated immediately prior to the Effective Time. If the Company's medical and/or dental benefit plans for Covered Employees are terminated prior to the end of a plan year, Covered Employees and their dependents who are then participating in a deductible-based medical and/or dental plan sponsored by the Company will be given credit for deductibles and eligible out-of-pocket expenses incurred towards deductibles and out-of-pocket maximums during the portion of the plan year preceding the termination date in a comparable deductible-based medical and/or dental plan of Parent or any of its Subsidiaries for the corresponding Parent benefit plan year.

(c) Except as otherwise provided in this Agreement, Parent and the Company shall honor, or cause to be honored, in accordance with their terms, all vested or accrued benefit obligations to, and contractual rights of, Covered Employees of the Company and its Subsidiaries, including any benefits or rights arising as a result of the Merger (either alone or in combination with any other event).

(d) Parent and the Company hereby agree that the Transactions shall constitute a "change in control" for purposes of any Benefit Arrangement for the Company to the extent such Benefit Arrangements include such term.

(e) The Company shall be permitted at any time prior to the Effective Time to pay annual bonuses based upon performance in its fiscal year ending January 31, 2006, in the ordinary course of business consistent with past practice in an amount currently accrued on the Company's consolidated balance sheet as of January 31, 2006. The Company has proposed to Merger Sub a bonus plan for the fiscal year ending January 31, 2007, a copy of which is set forth in Section 6.10(e) of the Disclosure Schedule, based upon targets and goals substantially similar to those established for its fiscal year ending January 31, 2007 in accordance with past practices. The adoption of such bonus plan for the year ended January 31, 2007 shall not be adopted without the consent of Parent and Merger Sub, but such consent shall not be unreasonably withheld. In granting such consent, Parent and Merger Sub shall be entitled in good faith to propose modifications to such plan that are reasonably intended to make such plan substantially similar in its purposes and effects to Parent's existing plans for its own employees. The Company will in good faith agree to changes proposed by Parent in such regards.

(f) The Covered Employees shall participate in Parent's bonus plans for similarly situated employees for the remainder of calendar year 2006 on the same terms and conditions as are Parent's other similarly situated employees. The Company's performance in respect of calculations made under the bonus plans for its fiscal year ending January 31, 2006, shall be calculated without taking into account any expenses or costs associated with or arising as a result of the Transactions or any nonrecurring charges that would not reasonably be expected to have been incurred had the Transactions not occurred and shall not be subject to reduction or negative discretion by the administrator of the bonus plan. If the Effective Time shall have occurred prior to the Company having paid bonuses for the fiscal year ending January 31, 2006, then Parent

shall pay such bonuses to the Covered Employees promptly following the Effective Time.

6.11 *Notification of Certain Matters.* Parent and the Company will give prompt notice to the other of any fact, event or circumstance known to it that would cause or constitute a breach of any of its representations, warranties, covenants or agreements contained herein that would be reasonably likely to cause a condition in Article VII not to be capable of being satisfied by the Termination Date.

6.12 *Parent Financing.* The Company hereby confirms and agrees that Wachovia Bank, National Association, its representatives and or affiliates (a) may assist, provide and/or arrange financing (the "*Financing*") in order for Parent to directly or indirectly acquire, or agree to acquire, offer, seek or propose to acquire directly or indirectly, ownership of the Company or its Subsidiaries or any securities issued by the Company or its Subsidiaries or any rights or options to acquire such ownership, all in accordance with the terms and conditions of this Agreement, and (b) following the Effective Time and the Financing, may exercise any rights or remedies available to it under the credit documentation or otherwise, and that such actions will not be deemed to be in violation of the standstill provisions of the Confidentiality Agreement.

ARTICLE VII CONDITIONS TO THE MERGER

7.01 *Conditions to Each Party's Obligation to Effect the Merger.* The respective obligation of each party to consummate the Merger is subject to the fulfillment or written waiver by the parties before the Effective Time of each of the following conditions:

(a) Shareholder Approval. The Company Shareholder Approval shall have been obtained.

(b) HSR Expiration. The applicable waiting period under the HSR Act shall have expired or been terminated.

(c) No Order. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any law, rule, regulation, injunction, order, decree or ruling (whether temporary, preliminary or permanent) which is then in effect and has the effect of making the Merger illegal or otherwise restricting, preventing or prohibiting consummation of the Merger.

7.02 *Conditions to the Company's Obligation.* The Company's obligation to consummate the Merger is also subject to the fulfillment or written waiver by the Company before the Effective Time of each of the following conditions:

(a) Parent's Representations and Warranties. The representations and warranties of Parent in this Agreement, including those relating to Merger Sub, shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such date (except to the extent such representations and warranties speak as of an earlier date, in which case such representations and warranties shall be true

and correct as of such date) without regard to the standard set forth in Section 5.02 of this Agreement, except where the aggregate effect of all such failures to be true and correct would not have a Material Adverse Effect on Parent; and the Company shall have received a certificate, dated the Closing Date, signed on behalf of Parent by the Chief Executive Officer or Chief Financial Officer of Parent to that effect.

(b) Performance of Parent's Obligations. Parent shall have performed in all material respects all obligations required to be performed by it under this Agreement at or before the Effective Time; and the Company shall have received a certificate, dated the Closing Date, signed on behalf of Parent by the Chief Executive Officer or Chief Financial Officer of Parent to that effect.

7.03 *Conditions to Parent's Obligation.* Parent's obligation to consummate the Merger is also subject to the fulfillment, or written waiver by Parent, before the Effective Time of each of the following conditions:

(a) The Company's Representations and Warranties. The representations and warranties of the Company in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such date (except to the extent such representations and warranties speak as of an earlier date, in which case such representations and warranties shall be true and correct as of such date) without regard to the standard set forth in Section 5.02 of this Agreement, except where the aggregate effect of all such failures to be true and correct would not have a Material Adverse Effect on the Company; and Parent shall have received a certificate, dated the Closing Date, signed on behalf of the Company by the Chief Executive Officer or Chief Financial Officer of the Company to that effect.

(b) Performance of the Company's Obligations. The Company shall have performed in all material respects all obligations required to be performed by it under this Agreement at or before the Effective Time; and Parent shall have received a certificate, dated the Closing Date, signed on behalf of the Company by the Chief Executive Officer or Chief Financial Officer of the Company to that effect.

(c) Governmental Consent. Any approval or consent of any Governmental Authority which is necessary for the Transactions to be consummated in accordance with the terms of the Agreement, or any relevant statutory, regulatory or other governmental waiting periods, whether domestic, foreign or supranational, shall have been obtained or be in full force and effect or shall have expired, except as would not be material and adverse to the Company and its Subsidiaries taken as a whole.

(d) No Material Adverse Change. No event, change, effect, condition, fact or circumstance shall have occurred following the date of this Agreement that, individually or in the aggregate, constitutes or would constitute a Material Adverse Change.

(e) No U.S. Governmental Litigation. There shall not have been instituted or be pending any action, proceeding, application, claim or counterclaim by any United States Federal Governmental Authority seeking to restrain or prohibit the consummation of the Merger.

ARTICLE VIII TERMINATION

8.01 *Termination.* This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time for any reason provided in paragraphs (a) through (f) below.

(a) By mutual written consent of each of Parent and the Company, notwithstanding any approval and adoption of this Agreement by the shareholders of the Company.

(b) By either Parent or the Company if any Governmental Authority shall have enacted, issued, promulgated, enforced or entered any injunction, order, decree or ruling (whether temporary, preliminary or permanent) which has become final and nonappealable and has the effect of making consummation of the Merger illegal or otherwise preventing or prohibiting consummation of the Merger; provided that the party seeking to terminate this Agreement shall have used its reasonable best efforts to remove or lift such injunction, order, decree or ruling.

(c) By either Parent or the Company (provided that the terminating party is not then in material breach of any representation, warranty, covenant or other agreement contained herein) if there shall have been a breach of any of the covenants or agreements or any of the representations or warranties set forth in this Agreement on the part of the other party such that if the non-breaching party is Merger Sub, Merger Sub would be entitled to not consummate the Merger pursuant to Section 7.02 (a) or (b) or if the non-breaching party is the Company, the Company would be entitled to not consummate the Merger pursuant to Section 7.03 (a) or (b), which breach has not been cured within the 20 days following written notice to the breaching party except for any breach which, by its nature or timing, cannot be cured.

(d) By Parent or the Company if the Effective Time has not occurred by the close of business on November 27, 2006 (the *"Termination Date"*), provided, however, that the right to terminate this Agreement under this Section 8.01(d) shall not be available to any party to the extent that such party's failure to comply with any provision of this Agreement, including without limitation Section 6.01, has resulted in the failure of any of the conditions set forth in Article VII hereto.

(e) By the Company if (i) the Company Board approves or recommends, or the Company enters into, a definitive agreement providing for an Acquisition Proposal, or (ii) the Company Board makes a Company Board Change of Recommendation, in each case only to the extent the Company has acted in accordance with, and has otherwise complied with the terms of, Section 6.06(b).

(f) By Parent if: (i) the Company violates in any material respect any of the provisions of Section 6.06 to the material detriment of Parent; (ii) the Company Board

recommends to the Company's shareholders any Acquisition Proposal or Superior Proposal; (iii) the Company enters into a definitive agreement providing for any Acquisition Proposal or Superior Proposal; or (iv) the Company Board or any committee thereof shall withdraw the approval of the Merger or this Agreement or effect a Company Board Change of Recommendation, or shall have approved and/or recommended, as applicable, an Acquisition Proposal.

8.02 *Effect of Termination.* In the event of the termination of this Agreement pursuant to Section 8.01, this Agreement shall forthwith become void and there shall be no liability on the part of any party hereto or any of its affiliates, directors, officers or stockholders or shareholders except that the Company or Parent may have liability or obligations as set forth in Section 8.03 and as set forth in or contemplated by Section 8.01 hereof. Notwithstanding the foregoing, nothing herein shall relieve the Company or Parent from liability for any willful breach hereof or willful misrepresentation herein.

8.03 Expenses.

(a) Except as otherwise specifically provided herein, all costs and expenses incurred in connection with this Agreement and the Transactions shall be paid by the party incurring such expenses, whether or not the Merger is consummated.

(b) In the event that this Agreement is terminated by (i) the Company pursuant to Section 8.01(e), (ii) the Parent pursuant to Section 8.01(f), (iii) the Parent pursuant to Section 8.01(c), or (iv) the Company or the Parent pursuant to Section 8.01(d) by reason of Section 7.01(a) not having been satisfied and in the case of such a termination pursuant to Section 8.01(c) or 8.01(e), (A) after the date of this Agreement and prior to in the case of clause (iii), the applicable breach giving rise to such termination right or in the case of clause (iv) the applicable Shareholders' Meeting, an Acquisition Proposal shall have been publicly announced or otherwise publicly communicated to the shareholders of the Company generally and in each case not withdrawn and (B) prior to the six month anniversary of such termination, (i) the Company shall enter into a definitive agreement with respect to such Acquisition Proposal or (ii) such Acquisition Proposal is consummated, then in the case of clauses (i) or (ii) of this Section 8.03(b) the Company shall not later than one Business Day following such termination, pay to Parent in immediately available funds an amount equal to \$37,340,000 (the "Termination Fee") and, in the case of clause (iv) of this Section 8.03(b) the Company shall pay to Parent in immediately available funds an amount equal to (x) in the case of clause (iii), the Termination Fee, or (y) in the case of clause (iv), the reasonable expenses actually incurred by Parent and Merger Sub in connection with the negotiation, preparation, execution and performance of this Agreement, in the cases of clauses (x) and (y), if and in the event of the consummation of such Acquisition Proposal no later than one Business Day following the consummation of such Acquisition Proposal. The Company acknowledges that the agreements contained in this Section 8.03(b) are an integral part of the transactions contemplated by this Agreement, and that, without these agreements, the Parent and the Merger Sub would not enter into this Agreement; accordingly, if the Company fails to pay the amount due pursuant to this Section 8.03(b), and, in order to obtain such payment, the Parent or the

Merger Sub commences a suit which results in a judgment against the Company for the fee set forth in this Section 8.03(b), the Company shall pay to the Parent or the Merger Sub, as the case may be, its costs and expenses (including attorneys' fees and expenses) in connection with such suit, together with interest on the amount of the fee at the prime rate in effect on the date such payment was required to be made. The Company shall pay such amount to Parent by wire transfer of same day funds to an account designated by Parent.

ARTICLE IX MISCELLANEOUS

9.01 *Survival.* The representations, warranties, agreements and covenants contained in this Agreement will not survive the Effective Time (other than Sections 2.04, 2.05, 2.06, 3.01 3.02, 3.03, 3.04, 6.09 and 6.10 and this Article IX).

9.02 *Waiver; Amendment.* At any time prior to the Effective Time, any provision of this Agreement may be (a) waived by the party benefited by the provision, but only in writing, or (b) amended or modified at any time, but only by a written agreement executed in the same manner as this Agreement, except to the extent that any such amendment would violate applicable law or require submission or resubmission of this Agreement or the Plan of Merger contained herein to the shareholders of the Company.

9.03 *Counterparts.* This Agreement may be executed in one or more counterparts, each of which will be deemed to constitute an original.

9.04 *Governing Law; Jurisdiction; Venue.* This Agreement is governed by, and will be interpreted in accordance with, the laws of the State of Texas applicable to contracts made and to be performed entirely within that State. Each of Parent, Merger Sub, and the Company hereby irrevocably and unconditionally; (i) consents to submit to the exclusive jurisdiction of the state and federal courts located in the State of Texas (*"Texas Courts"*) for any litigation arising out of or relating to this Agreement and the Transactions (and agrees not to commence any litigation relating thereto except in such courts), (ii) waives any objection to the laying of venue of any such litigation in the Texas Courts, and (iii) agrees not to plead or claim in any Texas Court that such litigation brought therein has been brought in any inconvenient forum.

9.05 *Notices.* All notices, requests and other communications given or made under this Agreement must be in writing and will be deemed given when personally delivered, facsimile transmitted (with confirmation) or mailed by registered or certified mail (return receipt requested) to the persons and addresses set forth below or such other place as such party may specify by notice.

If to the Company, to:

Stewart & Stevenson Services, Inc. 2707 North Loop West, Suite 800 Houston, Texas 77008 Attention: General Counsel Fax: (713) 868-2130

With a copy to:

Charles Henry Still Fulbright & Jaworski L.L.P. Fulbright Tower 1301 McKinney, Suite 5100 Houston, Texas 77010-3095 Fax: (713) 651-5246

If to Parent or Merger Sub, to:

Armor Holdings, Inc. 13386 International Parkway Jacksonville, Florida 32218 Attention: Robert R. Schiller Fax: (904) 741-5403

With copies to:

Kane Kessler, P.C. 1350 Avenue of the Americas, 26th Floor New York, New York 10019 Attention: Robert L. Lawrence, Esq. Fax: 212-245-3009

9.06 *Entire Understanding; No Third Party Beneficiaries.* This Agreement represents the entire understanding of Parent and the Company regarding the Transactions and supersedes any and all other oral or written Agreements previously made or purported to be made, other than the Confidentiality Agreement. No representation, warranty, inducement, promise, understanding or condition not set forth in this Agreement has been made or relied on by any party in entering into this Agreement. Except for Article II, and Section 6.09, which is intended to benefit the Indemnified Parties, and Section 6.10, which is intended to benefit Covered Employees and others, nothing expressed or implied in this Agreement is intended to confer any rights, remedies, obligations or liabilities upon any person other than Parent and the Company.

9.07 *Severability.* If any provision of this Agreement or the application thereof to any person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions, or the application of such provision to persons or circumstances other than those as to which it has been held invalid or unenforceable, will remain

in full force and effect and will in no way be affected, impaired or invalidated thereby, so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any party. Upon any such determination, the parties will negotiate in good faith in an effort to agree upon a suitable and equitable substitute provision to effect the original intent of the parties.

9.08 *Assignment; Successors.* No party may assign either this Agreement or any of its rights or interests, or delegate any of its duties hereunder, in whole or in part, without the prior written consent of the other parties. Any attempt to make any such assignment without such consent shall be null and void. Subject to the preceding sentences, this Agreement will be binding upon, inure to the benefit of and be enforceable by, the parties and their respective successors and permitted assigns.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized officers as of the day and year first above written.

	PARENT: ARMOR HOLDINGS, INC.	
	By:	/s/ PHILIP BARATELLI
	Name:	Philip Baratelli
	Title:	Corporate Controller, Treasurer & Secretary
	MERGER SUB: SANTANA ACQUISITION CORP.	
	By:	/s/ PHILIP BARATELLI
	Name:	Philip Baratelli
	Title:	Vice President and Secretary
	COMPANY: STEWART & STEVENSON SERVICES, INC.	
	By:	/s/ L. SCOTT BIAR
	Name:	L. Scott Biar
	Title:	Chief Financial Officer
A	\-4 4	

APPENDIX B

February 26, 2006

Board of Directors Stewart & Stevenson Services, Inc. 2707 North Loop West Houston, TX 77008

Members of the Board of Directors:

Stewart & Stevenson Services, Inc. (the "Company"), Armor Holdings, Inc. ("Parent") and Santana Acquisition Corp., a newly formed, wholly owned subsidiary of Parent ("MergerSub"), propose to enter into an Agreement and Plan of Merger (the "Agreement") pursuant to which MergerSub would be merged with the Company in a merger (the "Merger") in which each outstanding share of the Company's common stock, without par value, (the "Company Shares"), other than Company Shares held in treasury or held by Parent or any affiliate of Parent or as to which dissenter's rights have been perfected, would be converted into the right to receive \$35.00 per share in cash (the "Consideration").

You have asked us whether, in our opinion, the Consideration to be received by the holders of the Company Shares pursuant to the Merger is fair from a financial point of view to such holders.

In arriving at the opinion set forth below, we have, among other things:

(1)

Reviewed certain publicly available business and financial information relating to the Company that we deemed to be relevant;

(2)

Reviewed certain information, including financial forecasts, relating to the business, earnings, cash flow, assets, liabilities and prospects of the Company furnished to us by the Company;

(3)

Conducted discussions with members of senior management of the Company concerning the matters described in clauses 1 and 2 above;

(4)

Reviewed the market prices and valuation multiples for the Company Shares and compared them with those of certain publicly traded companies that we deemed to be relevant;

(5)

Reviewed the results of operations of the Company and compared them with those of certain publicly traded companies that we deemed to be relevant;

(6)

Compared the proposed financial terms of the Merger with the financial terms of certain other transactions that we deemed to be relevant;

(7)

Participated in certain discussions and negotiations among representatives of the Company and Parent and their financial and legal advisors;

(8)

Reviewed a draft dated February 26, 2006 of the Agreement; and

(9)

Reviewed such other financial studies and analyses and took into account such other matters as we deemed necessary, including our assessment of general economic, market and monetary conditions.

In preparing our opinion, we have assumed and relied on the accuracy and completeness of all information supplied or otherwise made available to us, discussed with or reviewed by or for us, or publicly available, and we have not assumed any responsibility for independently verifying such information or undertaken an independent evaluation or appraisal of any of the assets or liabilities of the Company or been furnished with any such evaluation or appraisal, nor have we evaluated the solvency or fair value of the Company under any state or federal laws relating to bankruptcy, insolvency or similar matters. In addition, we have not assumed any obligation to conduct any physical inspection of the properties or facilities of the Company. With respect to the financial forecast information furnished to or discussed with us by the Company, we have assumed that they have been reasonably prepared and reflect the best currently available estimates and judgment of the Company's management as to the expected future financial performance of the Company. We have also assumed that the final form of the Agreement will be substantially similar to the last draft reviewed by us.

Our opinion is necessarily based upon market, economic and other conditions as they exist and can be evaluated on, and on the information made available to us as of, the date hereof.

We are acting as financial advisor to the Company in connection with the Merger and will receive a fee from the Company for our services, a significant portion of which is contingent upon the consummation of the Merger. In addition, the Company has agreed to indemnify us for certain liabilities arising out of our engagement. We have, in the past, provided financial advisory and financing services to the Company, including financial advisory services to the Company in connection with the sale of its Power Products Division, and financial advisory and financing services to Parent and may continue to do so and have received, and may receive, fees for the rendering of such services. In addition, in the ordinary course of our business, we may actively trade the Company Shares and other securities of the Company, as well as securities of Parent for our own account and for the accounts of customers and, accordingly, may at any time hold a long or short position in such securities.

This opinion is for the use and benefit of the Board of Directors of the Company. Our opinion does not address the merits of the underlying decision by the Company to engage in the Merger and does not constitute a recommendation to any shareholder as to how such shareholder should vote on the proposed Merger or any matter related thereto. In addition, you have not asked us to address, and this opinion does not address, the fairness to, or any other consideration of, the holders of any class of securities, creditors or other constituencies of the Company, other than the holders of the Company Shares.

On the basis of and subject to the foregoing, we are of the opinion that, as of the date hereof, the Consideration to be received by the holders of the Company Shares pursuant to the Merger is fair from a financial point of view to the holders of such shares.

Very truly yours,

MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED

B-2

APPENDIX C

Art. 5.11. Rights of Dissenting Shareholders in the Event of Certain Corporate Actions

A. Any shareholder of a domestic corporation shall have the right to dissent from any of the following corporate actions:

(1) Any plan of merger to which the corporation is a party if shareholder approval is required by Article 5.03 or 5.16 of this Act and the shareholder holds shares of a class or series that was entitled to vote thereon as a class or otherwise;

(2) Any sale, lease, exchange or other disposition (not including any pledge, mortgage, deed of trust or trust indenture unless otherwise provided in the articles of incorporation) of all, or substantially all, the property and assets, with or without good will, of a corporation if special authorization of the shareholders is required by this Act and the shareholders hold shares of a class or series that was entitled to vote thereon as a class or otherwise;

(3) Any plan of exchange pursuant to Article 5.02 of this Act in which the shares of the corporation of the class or series held by the shareholder are to be acquired.

B. Notwithstanding the provisions of Section A of this Article, a shareholder shall not have the right to dissent from any plan of merger in which there is a single surviving or new domestic or foreign corporation, or from any plan of exchange, if:

(1) the shares, or depository receipts in respect of the shares, held by the shareholder are part of a class or series, shares, or depository receipts in respect of the shares, of which are on the record date fixed to determine the shareholders entitled to vote on the plan of merger or plan of exchange:

(a) listed on a national securities exchange;

(b) listed on the Nasdaq Stock Market (or successor quotation system) or designated as a national market security on an interdealer quotation system by the National Association of Securities Dealers, Inc., or successor entity; or

(c) held of record by not less than 2,000 holders;

(2) the shareholder is not required by the terms of the plan of merger or plan of exchange to accept for the shareholder's shares any consideration that is different than the consideration (other than cash in lieu of fractional shares that the shareholder would otherwise be entitled to receive) to be provided to any other holder of shares of the same class or series of shares held by such shareholder; and

(3) the shareholder is not required by the terms of the plan of merger or the plan of exchange to accept for the shareholder's shares any consideration other than:

(a) shares, or depository receipts in respect of the shares, of a domestic or foreign corporation that, immediately after the effective time of the merger or exchange, will be part of a class or series, shares, or depository receipts in respect of the shares, of which are:

(i) listed, or authorized for listing upon official notice of issuance, on a national securities exchange;

(ii) approved for quotation as a national market security on an interdealer quotation system by the National Association of Securities Dealers, Inc., or successor entity; or

(iii) held of record by not less than 2,000 holders;

(b) cash in lieu of fractional shares otherwise entitled to be received; or

(c) any combination of the securities and cash described in Subdivisions (a) and (b) of this subsection.

Art. 5.12. Procedure for Dissent by Shareholders as to Said Corporate Actions

A. Any shareholder of any domestic corporation who has the right to dissent from any of the corporate actions referred to in Article 5.11 of this Act may exercise that right to dissent only by complying with the following procedures:

(1) (a) With respect to proposed corporate action that is submitted to a vote of shareholders at a meeting, the shareholder shall file with the corporation, prior to the meeting, a written objection to the action, setting out that the shareholder's right to dissent will be exercised if the action is effective and giving the shareholder's address, to which notice thereof shall be delivered or mailed in that event. If the action is effected and the shareholder shall not have voted in favor of the action, the corporation, in the case of action other than a merger, or the surviving or new corporation (foreign or domestic) or other entity that is liable to discharge the shareholder's right of dissent, in the case of a merger, shall, within ten (10) days after the action is effected, deliver or mail to the shareholder written notice that the action has been effected, and the shareholder may, within ten (10) days from the delivery or mailing of the notice, make written demand on the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, for payment of the fair value of the shareholder's shares. The fair value of the shares shall be the value thereof as of the day immediately preceding the meeting, excluding any appreciation or depreciation in anticipation of the proposed action. In computing the fair value of the shares under this article, consideration must be given to the value of the corporation as a going concern without including in the computation of value any payment for a control premium or minority discount other than a discount attributable to the type of share held by the dissenting shareholder and any limitation placed on the rights and preferences of those shares. The demand shall state the number and class of the shares owned by the shareholder and the fair value of the shares as estimated by the shareholder. Any shareholder failing to make demand within the ten (10) day period shall be bound by the action.

(b) With respect to proposed corporate action that is approved pursuant to Section A of Article 9.10 of this Act, the corporation, in the case of action other than a merger, and the surviving or new corporation (foreign or domestic) or other entity that is liable to discharge the shareholder's right of dissent, in the case of a merger, shall, within ten (10) days after the date the action is effected, mail to each shareholder of record as of the effective date of the action notice of the fact and date of the action and that the shareholder may exercise the shareholder's right to dissent from the action. The notice shall be accompanied by a copy of this Article and any articles or documents filed by the corporation with the Secretary of State to effect the action. If the shareholder shall not have consented to the taking of the action, the shareholder may, within twenty (20) days after the mailing of the notice, make written demand on the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, for payment of the fair value of the shareholder's shares. The fair value of the shares shall be the value thereof as of the date the written consent authorizing the action was delivered to the corporation pursuant to Section A of Article 9.10 of this Act, excluding any appreciation or depreciation in anticipation of the shares as estimated by the shareholder. Any shareholder failing to make demand within the twenty (20) day period shall be bound by the action.

(2) Within twenty (20) days after receipt by the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, of a demand for payment made by a dissenting shareholder in accordance with Subsection (1) of this Section, the corporation (foreign

or domestic) or other entity shall deliver or mail to the shareholder a written notice that shall either set out that the corporation (foreign or domestic) or other entity accepts the amount claimed in the demand and agrees to pay that amount within ninety (90) days after the date on which the action was effected, and, in the case of shares represented by certificates, upon the surrender of the certificates duly endorsed, or shall contain an estimate by the corporation (foreign or domestic) or other entity of the fair value of the shares, together with an offer to pay the amount of that estimate within ninety (90) days after the date on which the action was effected, upon receipt of notice within sixty (60) days after that date from the shareholder that the shareholder agrees to accept that amount and, in the case of shares represented by certificates, upon the surrender of the certificates duly endorsed.

(3) If, within sixty (60) days after the date on which the corporate action was effected, the value of the shares is agreed upon between the shareholder and the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, payment for the shares shall be made within ninety (90) days after the date on which the action was effected and, in the case of shares represented by certificates, upon surrender of the certificates duly endorsed. Upon payment of the agreed value, the shareholder shall cease to have any interest in the shares or in the corporation.

B. If, within the period of sixty (60) days after the date on which the corporate action was effected, the shareholder and the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, do not so agree, then the shareholder or the corporation (foreign or domestic) or other entity may, within sixty (60) days after the expiration of the sixty (60) day period, file a petition in any court of competent jurisdiction in the county in which the principal office of the domestic corporation is located, asking for a finding and determination of the fair value of the shareholder's shares. Upon the filing of any such petition by the shareholder, service of a copy thereof shall be made upon the corporation (foreign or domestic) or other entity, which shall, within ten (10) days after service, file in the office of the clerk of the court in which the petition was filed a list containing the names and addresses of all shareholders of the domestic corporation (foreign or domestic) or other entity. If the petition shall be filed by the corporation (foreign or domestic) or other entity and to the shareholders named on the list at the addresses therein stated. The forms of the notices by mail shall be approved by the court. All shareholders named on the list at the addresses therein stated. The forms of the notices by mail shall be the final judgment of the court.

C. After the hearing of the petition, the court shall determine the shareholders who have complied with the provisions of this Article and have become entitled to the valuation of and payment for their shares, and shall appoint one or more qualified appraisers to determine that value. The appraisers shall have power to examine any of the books and records of the corporation the shares of which they are charged with the duty of valuing, and they shall make a determination of the fair value of the shares upon such investigation as to them may seem proper. The appraisers shall also afford a reasonable opportunity to the parties interested to submit to them pertinent evidence as to the value of the shares. The appraisers shall also have such power and authority as may be conferred on Masters in Chancery by the Rules of Civil Procedure or by the order of their appointment.

D. The appraisers shall determine the fair value of the shares of the shareholders adjudged by the court to be entitled to payment for their shares and shall file their report of that value in the office of the clerk of the court. Notice of the filing of the report shall be given by the clerk to the parties in interest. The report shall be subject to exceptions to be heard before the court both upon the law and the facts. The court shall by its judgment determine the fair value of the shares of the shareholders

entitled to payment for their shares and shall direct the payment of that value by the existing, surviving, or new corporation (foreign or domestic) or other entity, together with interest thereon, beginning 91 days after the date on which the applicable corporate action from which the shareholder elected to dissent was effected to the date of such judgment, to the shareholders entitled to payment. The judgment shall be payable to the holders of uncertificated shares immediately but to the holders of shares represented by certificates only upon, and simultaneously with, the surrender to the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, of duly endorsed certificates for those shares. Upon payment of the judgment, the dissenting shareholders shall cease to have any interest in those shares or in the corporation. The court shall allow the appraisers a reasonable fee as court costs, and all court costs shall be allotted between the parties in the manner that the court determines to be fair and equitable.

E. Shares acquired by the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, pursuant to the payment of the agreed value of the shares or pursuant to payment of the judgment entered for the value of the shares, as in this Article provided, shall, in the case of a merger, be treated as provided in the plan of merger and, in all other cases, may be held and disposed of by the corporation as in the case of other treasury shares.

F. The provisions of this Article shall not apply to a merger if, on the date of the filing of the articles of merger, the surviving corporation is the owner of all the outstanding shares of the other corporations, domestic or foreign, that are parties to the merger.

G. In the absence of fraud in the transaction, the remedy provided by this Article to a shareholder objecting to any corporate action referred to in Article 5.11 of this Act is the exclusive remedy for the recovery of the value of his shares or money damages to the shareholder with respect to the action. If the existing, surviving, or new corporation (foreign or domestic) or other entity, as the case may be, complies with the requirements of this Article, any shareholder who fails to comply with the requirements of this Article shall not be entitled to bring suit for the recovery of the value of his shares or money damages to the shareholder with respect to the action.

Art. 5.13. Provisions Affecting Remedies of Dissenting Shareholders

A. Any shareholder who has demanded payment for his shares in accordance with either Article 5.12 or 5.16 of this Act shall not thereafter be entitled to vote or exercise any other rights of a shareholder except the right to receive payment for his shares pursuant to the provisions of those articles and the right to maintain an appropriate action to obtain relief on the ground that the corporate action would be or was fraudulent, and the respective shares for which payment has been demanded shall not thereafter be considered outstanding for the purposes of any subsequent vote of shareholders.

B. Upon receiving a demand for payment from any dissenting shareholder, the corporation shall make an appropriate notation thereof in its shareholder records. Within twenty (20) days after demanding payment for his shares in accordance with either Article 5.12 or 5.16 of this Act, each holder of certificates representing shares so demanding payment shall submit such certificates to the corporation for notation thereon that such demand has been made. The failure of holders of certificated shares to do so shall, at the option of the corporation, terminate such shareholder's rights under Articles 5.12 and 5.16 of this Act unless a court of competent jurisdiction for good and sufficient cause shown shall otherwise direct. If uncertificated shares for which payment has been demanded or shares represented by a certificate on which notation has been so made shall be transferred, any new certificate issued therefor shall bear similar notation together with the name of the original dissenting holder of such shares and a transferee of such shares shall acquire by such transfer no rights in the corporation other than those which the original dissenting shareholder had after making demand for payment of the fair value thereof.

C. Any shareholder who has demanded payment for his shares in accordance with either Article 5.12 or 5.16 of this Act may withdraw such demand at any time before payment for his shares or before any petition has been filed pursuant to Article 5.12 or 5.16 of this Act asking for a finding and determination of the fair value of such shares, but no such demand may be withdrawn after such payment has been made or, unless the corporation shall consent thereto, after any such petition has been filed. If, however, such demand shall be withdrawn as hereinbefore provided, or if pursuant to Section B of this Article the corporation shall terminate the shareholder's rights under Article 5.12 or 5.16 of this Act, as the case may be, or if no petition asking for a finding and determination of fair value of such shares by a court shall have been filed within the time provided in Article 5.12 or 5.16 of this Act, as the case may be, or if after the hearing of a petition filed pursuant to Article 5.12 or 5.16, the court shall determine that such shareholder is not entitled to the relief provided by those articles, then, in any such case, such shareholder and all persons claiming under him shall be conclusively presumed to have approved and ratified the corporate action from which he dissented and shall be bound thereby, the right of such shareholder to be paid the fair value of his shares shall cease, and his status as a shareholder shall be restored without prejudice to any corporate proceedings which may have been taken during the interim, and such shareholder shall be entitled to receive any dividends or other distributions made to shareholders in the interim.

Front Side of Proxy

STEWART & STEVENSON, SERVICES, INC. SPECIAL MEETING TUESDAY, MAY 9, 2006

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS

The undersigned hereby appoints CARL B. KING and RITA M. SCHAULAT, or either of them, each with power to appoint his or her substitute, as proxies of the undersigned and authorizes them to represent and vote, as designated below, all the shares of Common Stock of Stewart & Stevenson Services, Inc. that the undersigned would be entitled to vote if personally present, and to act for the undersigned at the special meeting to be held at 10:00 a.m. on Tuesday, May 9, 2006 at the Sheraton Houston Brookhollow Hotel, 3000 North Loop West, Houston, Texas 77092, or any adjournment or postponement thereof.

This proxy revokes all prior proxies with respect to the special meeting. This proxy will be voted in the manner directed herein and in accordance with the accompanying Proxy Statement. Receipt of the Notice of Special Meeting and the related Proxy Statement is hereby acknowledged. If no direction is made, the undersigned grants the Proxies discretionary authority with respect to, and this proxy will be voted FOR, proposals 1 and 2 which are being proposed by the Board of Directors of Stewart & Stevenson Services, Inc. In their discretion, the Proxies are authorized to consider and vote upon such other matters as may properly come before the special meeting or any adjournment or postponement thereof.

(TO BE SIGNED ON THE REVERSE SIDE)

Back Side of Proxy

SIGN, DATE AND RETURN THIS PROXY IN THE ENCLOSED POSTAGE PAID ENVELOPE

The Board of Directors recommends you vote FOR the approval of proposals 1 and 2

1.

To approve and adopt the Agreement and Plan of Merger dated as of February 27, 2006, among Stewart & Stevenson Services, Inc., Armor Holdings, Inc. and Santana Acquisition Corp. and the merger contemplated by the merger agreement.

o FOR

o AGAINST

o ABSTAIN

2.

To approve the postponement or adjournment of the special meeting to a later date to solicit additional proxies in favor of the approval and adoption of the merger agreement and the merger if there are not sufficient votes for approval and adoption of the merger agreement and the merger.

o FOR o AGAINST o ABSTAIN In accordance with their discretion, to consider and vote upon such other matters as may properly come before the special meeting or any adjournment or postponement of the special meeting.

The signature must be that of the shareholder himself, herself or itself. If shares are held jointly, each shareholder named should sign. If a signer is a corporation, please sign the full corporate name by duly authorized officer. If the signer is a partnership, please sign the full partnership name by authorized person. Executors, administrators, trustees, guardians, attorneys-in-fact, etc., should so indicate when signing.

Date

Share Owner sign here

Co-Owner sign here